COLLEEN CHAWLA, Director



July 27, 2018

Valero Energy Corporation One Valero Way San Antonio, Texas 78249 Attention: Mr. C. Shay Wideman (Sent via e-mail to: <u>Shay.Wideman@valero.com</u>)

Subject: Technical Report Request for Fuel Leak Case No. RO0003115 and GeoTracker Global ID T10000004796, Stop N Go Gas Station, 7701 Bancroft Avenue, Oakland, CA 94605

Dear Mr. Wideman:

Alameda County Department of Environmental Health (ACDEH) staff has reviewed the case file including Valero Energy Corporation's (Valero's) March 15, 2018 letter (Attachment A) which asserts that Valero never owned or operated the subject property and consequently should not be named as a Responsible Party (RP).

ACDEH issued a Notice of Responsibility (NOR) on September 26, 2016 (Attachment B) which named Valero Energy Corporation as an RP. ACDEH names an RP as defined under 23 California Code of Regulation (C.C.R.) Sec. 2720 and Section 2720 defines an RP four ways.

A RP can be:

- 1. "Any person who owns or operates an underground storage tank used for the storage of any hazardous substance."
- 2. "In the case of any underground storage tank no longer in use, any person who owned or operated the underground storage tank immediately before the discontinuation of its use."
- 3. "Any owner of property where an unauthorized release of a hazardous substance from an underground storage tank has occurred."
- 4. "Any person who had or has control over an underground storage tank at the time of or following an unauthorized release of a hazardous substance."

ACDEH's rationale for naming Valero was provided in the September 26, 2016 NOR *Responsible Party Identification* of *Attachment A – Responsible Party Data Sheet,* which states:

National Convenience Stores Incorporated (operating under the name Stop N Go) purchased or acquired the property in December 13, 1973. The National Convenience Stores Incorporated was acquired by Diamond Shamrock in 1995; Diamond Shamrock was acquired in 1996 by Ultramar and was renamed Ultramar Diamond Shamrock. Ultramar Diamond Shamrock was acquired by Valero Energy Corporation in 2001. Valero Energy Corporation, as a successor to National Convenience Stores Incorporated, is a responsible party because they owned or operated the USTs used for the storage of any hazardous substance (Definition 1) and they were the property owner where an unauthorized release of a hazardous substance from the USTs occurred (Definition 3).

Despite the Separation and Distribution Agreement by and between Valero Energy Corporation and CST Brands, Inc. dated April 29, 2013 (Attachment C) submitted to ACDEH, ACDEH's evaluation of RP status only considers 23 California Code of Regulation (C.C.R.) Sec. 2720 in the naming RPs.

ACDEH notes that the NOR was dated September 26, 2016, the date of Valero's letter disputing the RP status was dated March 15, 2018, and the NOR states that *Any action or inaction by this local agency associated with corrective action, including responsible party identification, is subject to petition to the State Water*

Mr. C. Shay Wideman RO00003115 July 27, 2018, Page 2

Resources Control Board. Petitions must be filed within 30 days from the date of the action/inaction. The deadline for filing a petition exceeds 30 days; however, you are welcome to dispute your RP status by following this link to file a petition with the State Water Resources Control Board:

https://www.waterboards.ca.gov/public notices/petitions/water quality/wqpetition instr.shtml

Thank you for your cooperation. Should you have any questions or concerns regarding this correspondence or your case, please send me an e-mail message at <u>karel.detterman@acgov.org</u> or call me at (510) 567-6708.

Sincerely,

Karel Detterman, P.G. Senior Hazardous Materials Specialist

Enclosures: Attachment A – Valero Energy Corporation Correspondence dated March 15, 2018

Attachment B - Notice of Responsibility dated September 26, 2016

Attachment C - Separation and Distribution Agreement by and between Valero Energy Corporation and CST Brands, Inc. Execution Version, dated April 29, 2013

Attachment 1 – Responsible Party (ies) Legal Requirements/Obligations Electronic Report Upload (ftp) Instructions

 cc: Victoria Vela, Bancroft and 77th, LLC, Sent via e-mail to: <u>v.vela@pwslaundry.com</u>) Carryl MacLeod, Chevron Environmental Management Company, Sent via e-mail to: <u>CMacleod@chevron.com</u> Gerard Sonnier, National Convenience Stores Inc. c/o CST Brands, Inc., Sent via e-mail to: <u>Gerard.Sonnier@cstbrands.com</u> Kee S. and Ip C. Lee, 4461 Cerritos Avenue, Cypress, CA 90630-4217 Bank of America NT&SA, c/o National Convenience Stores, Inc., 100 N. Tryon Street, Charlotte, North Carolina, 28202 Bancroft Associates, LLC, Justin Smith, Address Unknown Justin E. Smith and William C. Randall, Address Unknown Cathleen Maynard Trust, Address Unknown Earl L. & Marjorie E. Lupton, Jr. Trust, Address Unknown

Kathleen Clack, Law Offices (sent via e-mail to: <u>kpc@clacklaw.com</u>) Brad Steinberg, PWS Laundry, (sent via e-mail to: <u>bsteinberg@pwslaundry.com</u>) Daniel A. Robinson, Esq., CST Brands, Inc. (Sent via e-mail to: <u>daniel.robinson@cstbrands.com</u>) Sharon Vasquez, Chevron, (sent via e-mail to: <u>sharonvasquez@chevron.com</u>) Robert Goodman, (sent via e-mail to: <u>rgoodman@rjo.com</u>) Brian Silva, GHD, (sent via e-mail to: <u>brian.silva@ghd.com</u>) Kerri Barsh, Greenberg Traurig, (sent via e-mail to: <u>barshk@gtlaw.com</u>) John Chibbaro, Greenberg Traurig, (sent via e-mail to: <u>chibbaroj@gtlaw.com</u>)

Dilan Roe, ACDEH, (sent via e-mail to: <u>dilan.roe@acgov.org</u>) Karel Detterman, ACDEH, (sent via e-mail to: <u>karel.detterman@acgov.org</u>) Paresh Khatri, ACDEH, (sent via e-mail to: <u>paresh.khatri@acgov.org</u>) Case Electronic File, GeoTracker

ATTACHMENT A



March 15, 2018

Karel Detterman Alameda County Health Care Services Agency 1131 Harbor Bay Parkway, Suite 250 Alameda, CA 94502

Re: Response to letter dated January 30, 2018 asserting Valero Energy Corporation as RP for Case No. RO0003115 and GeoTracker Global ID T10000004796, Stop N Go Station, 7701 Bancroft Avenue, Oakland CA 94605

Ms. Detterman;

On April 29, 2013, Valero Energy Corporation and CST Brands, Inc. executed a Separation and Distribution Agreement (see attached) whereby retail Assets and Assumptions of Liability related to National Convenience Stores Incorporated were transferred to CST Brands, Inc. ("CST"). On June 28th, 2017 CST Brands, Inc. was acquired by Alimentation Couche-Tard.

Please be advised that Valero Energy Corporation has never owned or operated the subject property. Therefore, Valero demands that the Alameda County Health Care Services Agency (ACHCSA) immediately retract their position and issue a letter of correction.

Sincerely,

C. Shay Wideman Director – Environmental Liability and Remediation Management

ATTACHMENT B

ALAMEDA COUNTY HEALTH CARE SERVICES



AGENC)

REBECCA GEBHART, Interim Director

ENVIRONMENTAL HEALTH DEPARTMENT ENVIRONMENTAL PROTECTION 1131 Harbor Bay Parkway, Suite 250 Alameda, CA 94502-6577

September 26, 2016

Bancroft and 77th, LLC Attention: Ms. Victoria Vela 6500 Flotilla Avenue Los Angeles, CA 90040 (Sent via e-mail to: <u>v.vela@pwslaundry.com</u>)

Bancroft Associates, LLC Attention: Mr. Justin Smith Address Unknown

Cathleen Maynard Trust Address Unknown

Kee S. & Ip C. Lee 4461 Cerritos Avenue Cypress, CA 90630-4217

National Convenience Stores Inc. c/o CST Brands, Inc. Attention: Mr. Gerard Sonnier One Valero Way Building D, Suite 200 San Antonio, Texas 78249 Ms. Carryl MacLeod Chevron Environmental Management Company 6101 Bollinger Canyon Road San Ramon, CA 94583-2324 (Sent via e-mail to: <u>CMacleod@chevron.com</u>)

Justin E. Smith and William C. Randall Address Unknown

Earl L. & Marjorie E. Lupton, Jr. Trust Address Unknown

Bank of America NT&SA c/o National Convenience Stores, Inc. 100 N. Tryon Street Charlotte, North Carolina, 28202

Valero Energy Corporation One Valero Way San Antonio, Texas 78249

Subject: Notice of Responsibility Update for Fuel Leak Case No. RO0003115 and GeoTracker Global ID T10000004796, Stop N Go Gas Station, 7701 Bancroft Avenue, Oakland, CA 94605

Dear Ladies and Gentlemen:

In a Notice of Responsibility (NOR) dated February 6, 2014 and updated on August 21, 2015, numerous parties were notified that the above referenced site had been placed in the Local Oversight Program (LOP) and that they had been named as a Responsible Parties for the fuel leak case. An additional party has been named as a Responsible Parties for the fuel leak case have been updated in the attached updated NOR as defined under 23 C.C.R Sec. 2720. Please see Attachment A – Responsible Parties Data Sheet, which identifies all Responsible Parties and provides background on the unauthorized release and Responsible Party Identification.

Should you have any questions, please contact me at (510) 567-6708 or send me an e-mail message at <u>karel.detterman@acgov.org</u>.

Sincerely,

Karel Detterman, P.G. Hazardous Materials Specialist

Digitally signed by Karel Detterman DN: cn=Karel Detterman, o, ou, email=karel.detterman@acgov.org, c=US Date: 2016.09.22 15:59:05 -07'00' Ladies and Gentlemen RO00003115 September 26, 2016 Page 2

Enclosures: Attachment 1 – Responsible Party (ies) Legal Requirements / Obligations Electronic Report Upload (ftp) Instructions

Attachment A – Responsible Parties Data Sheet Notice of Responsibility

cc: Kathleen Clack, Law Offices (sent via e-mail to: <u>kpc@clacklaw.com</u>) Dilan Roe, ACDEH (sent via electronic mail to: <u>dilan.roe@acgov.org</u>) Karel Detterman, ACDEH, (sent via electronic mail to <u>karel.detterman@acgov.org</u>) Case Electronic File, GeoTracker



Certified Mail #: 7011 3500 0003 1848 1486

September 26, 2016

NOTICE OF RESPONSIBILITY

Site Name & Address: **STOP N GO GAS STATION**

7701 BANCROFT AVENUE **OAKLAND, CA 94505**

Local ID: RO0003115 Related ID: NA RWQCB ID: NA Global ID:

T1000004796

Responsible Party: VALERO ENERGY CORPORATION **ONE VALERO WAY** SAN ANTONIO, TEXAS 78249

Date First Reported:	11/13/2012
Substance: 12034	4, 12035, 02036, 8006619 Multiple Releases
Funding for Oversight: Multiple RPs?: Yes	LOPS - LOP State Fund

Pursuant to sections 25297.1 and 25297.15 of the Health and Safety Code, you are hereby notified that the above site has been placed in the Local Oversight Program and the individual(s) or entity(ies) shown above, or on the attached list, has (have) been identified as the party(ies) responsible for investigation and cleanup of the above site. Section 25297.15 further requires the primary or active Responsible Party to notify all current record owners of fee title before the local agency considers cleanup or site closure proposals or issues a closure letter. For purposes of implementing section 25297.15, this agency has identified VALERO ENERGY CORPORATION as the primary or active Responsible Party. It is the responsibility of the primary or active Responsible Party to submit a letter to this agency, within 20 calendar days of receipt of this notice that identifies all current record owners of fee title. It is also the responsibility of the primary or active Responsible Party to certify to the local agency that the required notifications have been made at the time a cleanup or site closure proposal is made or before the local agency makes a determination that no further action is required. If property ownership changes in the future, you must notify this local agency within 20 calendar days from when you are informed of the change.

Any action or inaction by this local agency associated with corrective action, including responsible party identification, is subject to petition to the State Water Resources Control Board. Petitions must be filed within 30 days from the date of the action/inaction. To obtain petition procedures, please FAX your request to the State Water Board at (916) 341-5808 or telephone (916) 341-5752.

Pursuant to section 25296.10(c)(6) of the Health and Safety Code, a responsible party may request the designation of an administering agency when required to conduct corrective action. Please contact this office for further information about the designation process.

Please contact your caseworker KAREL DETTERMAN at this office at (510) 567-6708 if you have questions regarding your site.

- Date: 09-26-2014

RONALD BROWDER, Director Contract Project Director

Action: Update ADD Reason:

Attachment A: Responsible Parties Data Sheet

cc: Cindy Davis, SWRCB (email: cindy.davis@waterboards.ca.gov) | Dilan Roe, ACDEH (email: dilan.roe@acgov.org), File

ALAMEDA COUNTY ENVIRONMENTAL HEALTH LUFT LOCAL OVERSIGHT PROGRAM

ATTACHMENT A - RESPONSIBLE PARTIES DATA SHEET

September 26, 2016

Site Name & Address: STOP N GO GAS STATION 7701 BANCROFT AVE OAKLAND, CA 94605 Local ID: RO0003115 Related ID: NA RWQCB ID: NA Global ID: T10000004796

All Responsible Parties

RP has been named a Primary RP - c/o JUSTIN SMITH BANCROFT ASSOCIATES, LLC ADDRESS UNKNOWN | No Phone Number Listed

RP has been named a Primary RP -CATHLEEN MAYNARD TRUST ADDRESS UNKNOWN | No Phone Number Listed

RP has been named a Primary RP - KEE S. & IP C. LEE 4461 CERRITOS AVENUE | CYPRESS, CA 90630--421 | No Phone Number Listed

RP has been named a Primary RP - CARRYL MACLEOD

CHEVRON ENVIRONMENTAL MANAGEMENT COMPANY 6101 BOLLINGER CANYON ROAD, ROOM 5323 | SAN RAMON, CA 94583-232 | No Phone Number Listed

RP has been named a Primary RP - JUSTIN E. SMITH & WILLIAM C. RANDALL ADDRESS UNKNOWN | No Phone Number Listed

RP has been named a Primary RP -EARL L & MARJORIE E LUPTON, JR TRUST ADDRESS UNKNOWN | No Phone Number Listed

RP has been named a Primary RP -BANK OF AMERICA 100 N. TRYON ST | CHARLOTTE, NORTH CAROLINA 28202 | No Phone Number Listed

RP has been named a Primary RP – VICTORIA VELA BANCROFT AND 77TH, LLC 6500 FLOTILLA AVE | LOS ANGELES, CA 90040 | No Phone Number Listed

RP has been named a Primary RP – GERARD SONNIER NATIONAL CONVENIENCE STORES INCORPORATED C/O CST BRANDS, INC. ONE VALERO WAY BUILDING D, SUITE 200 | SAN ANTONIO, TX 78249 | No Phone Number Listed

RP has been named a Primary RP – VALERO ENERGY CORPORATION ONE VALERO WAY | SAN ANTONIO, TX 78249 | No Phone Number Listed

ATTACHMENT A - RESPONSIBLE PARTIES DATA SHEET (Continued) September 26, 2016

Responsible Party Identification Background

Alameda County Environmental Health (ACEH) names a "Responsible Party," as defined under 23 C.C.R Sec. 2720. Section 2720 defines a responsible party four ways. An RP can be:

- 1. "Any person who owns or operates an underground storage tank used for the storage of any hazardous substance."
- 2. "In the case of any underground storage tank no longer in use, any person who owned or operated the underground storage tank immediately before the discontinuation of its use."
- 3. "Any owner of property where an unauthorized release of a hazardous substance from an underground storage tank has occurred."
- 4. "Any person who had or has control over an underground storage tank at the time of or following an unauthorized release of a hazardous substance."

ATTACHMENT A - RESPONSIBLE PARTIES DATA SHEET (Continued)

September 26, 2016

Responsible Party Identification

Existence of Unauthorized Release

In January 2012 seven direct-push soil borings were drilled onsite. Concentrations of up to 110 milligrams per kilogram (mg/kg) Total Petroleum Hydrocarbons as Gasoline (TPHG), 1,500 mg/kg TPH as diesel, and 4,500 mg/kg TPH as motor oil were documented in soil samples. Concentrations of up to 150 micrograms per liter (ug/l) TPH diesel and 370 ug/l TPH motor oil were documented in grab groundwater samples. These data indicate that an unauthorized release from the underground storage tanks (USTs) had occurred at the site.

Responsible Party Identification

Gulf Oil Corporation purchased or acquired the property January 21, 1962. The Gulf Oil Corporation was succeeded by Standard Oil of California, and then Chevron. Chevron, as a successor to the Gulf Oil Corporation, is a responsible party because they owned or operated the USTs used for the storage of any hazardous substance (Definition 1), in the case of any USTs no longer in use, they owned and operated the USTs immediately before the discontinuation of their use (Definition 2), and they were the property owner where an unauthorized release of a hazardous substance from the USTs occurred (Definition 3).

National Convenience Stores Incorporated (operating under the name Stop N Go) purchased or acquired the property in December 13, 1973. The National Convenience Stores Incorporated was acquired by Diamond Shamrock in 1995; Diamond Shamrock was acquired in 1996 by Ultramar and was renamed Ultramar Diamond Shamrock. Ultramar Diamond Shamrock was acquired by Valero Energy Corporation in 2001. Valero Energy Corporation, as a successor to National Convenience Stores Incorporated, is a responsible party because they owned or operated the USTs used for the storage of any hazardous substance (Definition 1) and they were the property owner where an unauthorized release of a hazardous substance from the USTs occurred (Definition 3).

Bank of America NT&SA in care of National Convenience Stores Incorporated purchased or acquired the property December 30, 1975. Bank of America NT&SA in care of National Convenience Stores Incorporated is a responsible party for the site because they owned or operated the USTs used for the storage of any hazardous substance (Definition 1), in the case of any USTs no longer in use, they owned the USTs immediately before the discontinuation of their use (Definition 2), and they were the property owner where an unauthorized release of a hazardous substance from the USTs occurred (Definition 3).

Earl L. and Marjorie E. Lupton and Kee S. and Ip C. Lee purchased or acquired the property April 26, 1989. Earl L. and Marjorie E. Lupton and Kee S. and Ip C. Lee are responsible parties for the site because they owned the property where an unauthorized release of a hazardous substance from a UST occurred (Definition 3).

The Earl L. and Marjorie E. Lupton Trust purchased or acquired the property November 12, 1996. The Earl L. and Marjorie E. Lupton Trust are responsible parties for the site because they owned the property where an unauthorized release of a hazardous substance from a UST occurred (Definition 3).

The Cathleen Maynard Trust and Justin Smith & Bill Randall purchased or acquired the property on November 21, 2001. The Cathleen Maynard Trust and Justin Smith & Bill Randall are responsible parties for the site because they owned the property where an unauthorized release of a hazardous substance from a UST occurred (Definition 3).

Justin E. Smith & William C. Randall purchased or acquired the property in September 30, 2005. Justin E. Smith & William C. Randall are responsible parties for the site because they owned the property where an unauthorized release of a hazardous substance from a UST occurred (Definition 3).

ATTACHMENT A - RESPONSIBLE PARTIES DATA SHEET (Continued) September 26, 2016

Responsible Party Identification (continued):

Bancroft Associates LLC in care of Justin Smith purchased or acquired the property in November 7, 2005. Bancroft Associates LLC is a responsible party for the site because it owned the property where an unauthorized release of a hazardous substance from a UST occurred (Definition 3).

Tamalpais Bank then Union Bank held the property in foreclosure between April 14, 2009 and May 4, 2015.

Bancroft and 77th, LLC, purchased or acquired the property on May 4, 2015. Bancroft and 77th, LLC is a responsible party for the site because it owns the property where an unauthorized release of a hazardous substance from a UST occurred (Definition 3).

Attachment 1

Responsible Party(ies) Legal Requirements / Obligations

REPORT REQUESTS

These reports are being requested pursuant to California Health and Safety Code Section 25296.10. 23 CCR Sections 2652 through 2654, and 2721 through 2728 outline the responsibilities of a responsible party in response to an unauthorized release from a petroleum UST system, and require your compliance with this request.

ELECTRONIC SUBMITTAL OF REPORTS

ACEH's Environmental Cleanup Oversight Programs (LOP and SLIC) require submission of reports in electronic form. The electronic copy replaces paper copies and is expected to be used for all public information requests, regulatory review, and compliance/enforcement activities. Instructions for submission of electronic documents to the Alameda County Environmental Cleanup Oversight Program FTP site are provided on the attached "Electronic Report Upload Instructions." Submission of reports to the Alameda County FTP site is an addition to existing requirements for electronic submittal of information to the State Water Resources Control Board (SWRCB) GeoTracker website. In September 2004, the SWRCB adopted regulations that require electronic submittal of information for all groundwater cleanup programs. For several years, responsible parties for cleanup of leaks from underground storage tanks (USTs) have been required to submit groundwater analytical data, surveyed locations of monitoring wells, and other data to the GeoTracker database over the Internet. Beginning July 1, 2005, these same reporting requirements were added to Spills, Leaks, Investigations, and Cleanup (SLIC) sites. Beginning July 1, 2005, electronic submittal of a complete copy of all reports for all sites is required in GeoTracker (in PDF format). Please visit the SWRCB website for more information on these requirements (http://www.waterboards.ca.gov/water issues/programs/ust/electronic submittal/).

PERJURY STATEMENT

All work plans, technical reports, or technical documents submitted to ACEH must be accompanied by a cover letter from the responsible party that states, at a minimum, the following: "I declare, under penalty of perjury, that the information and/or recommendations contained in the attached document or report is true and correct to the best of my knowledge." This letter must be signed by an officer or legally authorized representative of your company. Please include a cover letter satisfying these requirements with all future reports and technical documents submitted for this fuel leak case.

PROFESSIONAL CERTIFICATION & CONCLUSIONS/RECOMMENDATIONS

The California Business and Professions Code (Sections 6735, 6835, and 7835.1) requires that work plans and technical or implementation reports containing geologic or engineering evaluations and/or judgments be performed under the direction of an appropriately registered or certified professional. For your submittal to be considered a valid technical report, you are to present site specific data, data interpretations, and recommendations prepared by an appropriately licensed professional and include the professional registration stamp, signature, and statement of professional certification. Please ensure all that all technical reports submitted for this fuel leak case meet this requirement.

UNDERGROUND STORAGE TANK CLEANUP FUND

Please note that delays in investigation, later reports, or enforcement actions may result in your becoming ineligible to receive grant money from the state's Underground Storage Tank Cleanup Fund (Senate Bill 2004) to reimburse you for the cost of cleanup.

AGENCY OVERSIGHT

If it appears as though significant delays are occurring or reports are not submitted as requested, we will consider referring your case to the Regional Board or other appropriate agency, including the County District Attorney, for possible enforcement actions. California Health and Safety Code, Section 25299.76 authorizes enforcement including administrative action or monetary penalties of up to \$10,000 per day for each day of violation.

Alameda County Environmental Cleanup	REVISION DATE: May 15, 2014
Oversight Programs	ISSUE DATE: July 5, 2005
(LOP and SLIC)	PREVIOUS REVISIONS: October 31, 2005; December 16, 2005; March 27, 2009; July 8, 2010, July 25, 2010
SECTION: Miscellaneous Administrative Topics & Procedures	SUBJECT: Electronic Report Upload (ftp) Instructions

The Alameda County Environmental Cleanup Oversight Programs (LOP and SLIC) require submission of all reports in electronic form to the county's ftp site. Paper copies of reports will no longer be accepted. The electronic copy replaces the paper copy and will be used for all public information requests, regulatory review, and compliance/enforcement activities.

REQUIREMENTS

- Please <u>do not</u> submit reports as attachments to electronic mail.
- Entire report including cover letter must be submitted to the ftp site as a single portable document format (PDF) with no password protection.
- It is preferable that reports be converted to PDF format from their original format, (e.g., Microsoft Word) rather than scanned.
- Signature pages and perjury statements must be included and have either original or electronic signature.
- Do not password protect the document. Once indexed and inserted into the correct electronic case file, the document will be secured in compliance with the County's current security standards and a password. Documents with password protection will not be accepted.
- Each page in the PDF document should be rotated in the direction that will make it easiest to read on a computer monitor.
 Benots must be paged and eaved using the following and in the direction that will make it easiest to read on a computer
- Reports must be named and saved using the following naming convention:

RO#_Report Name_Year-Month-Date (e.g., RO#5555_WorkPlan_2005-06-14)

Submission Instructions

- 1) Obtain User Name and Password
 - a) Contact the Alameda County Environmental Health Department to obtain a User Name and Password to upload files to the ftp site.
 - i) Send an e-mail to <u>deh.loptoxic@acgov.org</u>
 - b) In the subject line of your request, be sure to include "ftp PASSWORD REQUEST" and in the body of your request, include the Contact Information, Site Addresses, and the Case Numbers (RO# available in Geotracker) you will be posting for.
- 2) Upload Files to the ftp Site
 - a) Using Internet Explorer (IE4+), go to ftp://alcoftp1.acgov.org
 - (i) Note: Netscape, Safari, and Firefox browsers will not open the FTP site as they are NOT being supported at this time.
 - b) Click on Page located on the Command bar on upper right side of window, and then scroll down to Open FTP Site in Windows Explorer.
 - c) Enter your User Name and Password. (Note: Both are Case Sensitive.)
 - d) Open "My Computer" on your computer and navigate to the file(s) you wish to upload to the ftp site.
 - e) With both "My Computer" and the ftp site open in separate windows, drag and drop the file(s) from "My Computer" to the ftp window.
- 3) Send E-mail Notifications to the Environmental Cleanup Oversight Programs
 - a) Send email to deh.loptoxic@acgov.org notify us that you have placed a report on our ftp site.
 - b) Copy your Caseworker on the e-mail. Your Caseworker's e-mail address is the entire first name then a period and entire last name @acgov.org. (e.g., firstname.lastname@acgov.org)
 - c) The subject line of the e-mail must start with the RO# followed by **Report Upload**. (e.g., Subject: RO1234 Report Upload) If site is a new case without an RO#, use the street address instead.
 - d) If your document meets the above requirements and you follow the submission instructions, you will receive a notification by email indicating that your document was successfully uploaded to the ftp site.

ATTACHMENT C

EXECUTION VERSION

SEPARATION AND DISTRIBUTION AGREEMENT

BY AND BETWEEN

VALERO ENERGY CORPORATION

AND

CST BRANDS, INC.

DATED AS OF APRIL 29, 2013

TABLE OF CONTENTS

ARTICI	E I DEFINITIONS	2
ARTICI	E II THE SEPARATION	. 12
2.1	Transfer of Assets and Assumption of Liabilities	12
2.2	Corner Store Assets	
2.3	Corner Store Liabilities	15
2.4	Transfer of Excluded Assets; Assumption of Excluded Liabilities	17
2.5	Approvals and Notifications	17
2.6	Novation of Corner Store Liabilities	19
2.7	Novation of Excluded Liabilities	20
2.8	Termination of Agreements	20
2.9	Treatment of Shared Contracts	21
2.10	Bank Accounts; Cash Balances	22
2.11	Other Ancillary Agreements	23
2.12	Disclaimer of Representations and Warranties	23
ARTICI	E III THE DISTRIBUTION	24
3.1	Actions Prior to the Distribution	24
3.2	Effecting the Distribution	26
3.3	Conditions to the Distribution	27
3.4	Sole Discretion	28
ARTICL	E IV DISPUTE RESOLUTION	28
4.1	General Provisions	28
4.2	Consideration by Senior Executives	29
4.3	Arbitration	29
ARTICI	E V MUTUAL RELEASES; INDEMNIFICATION	31
5.1	Release of Pre-Distribution Claims	31
5.2	Indemnification by Corner Store	33
5.3	Indemnification by Valero	34
5.4	Indemnification Obligations Net of Insurance Proceeds	35
5.5	Procedures for Indemnification of Third-Party Claims	36
5.6	Additional Matters	38
5.7	Remedies Cumulative	40
5.8	Survival of Indemnities	40
5.9	Guarantees. Letters of Credit and Other Obligations	40

5.10	Right of Contribution	. 41
5.11	No Impact on Third Parties	. 41
5.12	No Cross-Claims or Third-Party Claims	. 41
5.13	Severability	. 42
5.14	Ancillary Agreements	. 42
ARTICL	E VI INSURANCE MATTERS	. 42
6.1	Insurance Matters	. 42
ARTICL	E VII EXCHANGE OF INFORMATION; CONFIDENTIALITY	. 45
7.1	Agreement for Exchange of Information	. 45
7.2	Ownership of Information	. 45
7.3	Compensation for Providing Information	. 45
7.4	Record Retention	. 46
7.5	Limitations of Liability	. 46
7.6	Other Agreements Providing for Exchange of Information	. 46
7.7	Production of Witnesses; Records; Cooperation	. 47
7.8	Privileged Matters	. 47
7.9	Confidentiality	. 49
7.10	Protective Arrangements	. 50
ARTICL	E VIII FURTHER ASSURANCES AND ADDITIONAL COVENANTS	50
ARTICL	E VIII FURTHER ASSURANCES AND ADDITIONAL COVENANTS	
		50
8.1	Further Assurances	50 51
8.1 8.2 8.3	Further Assurances Performance	50 51 51
8.1 8.2 8.3	Further Assurances Performance	50 51 51 51
8.1 8.2 8.3 ARTICL 9.1	Further Assurances Performance Order of Precedence E IX TERMINATION	50 51 51 52 52
8.1 8.2 8.3 ARTICL 9.1	Further Assurances Performance Order of Precedence E IX TERMINATION Termination	50 51 51 52 52 52
8.1 8.2 8.3 ARTICL 9.1 ARTICL	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS	50 51 51 52 52 52 52
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS Counterparts; Entire Agreement; Corporate Power	50 51 52 52 52 52 52 52
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1 10.2	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS Counterparts; Entire Agreement; Corporate Power Governing Law	. 50 . 51 . 51 . 52 . 52 . 52 . 52 . 52 . 53 . 53
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1 10.2 10.3	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS Counterparts; Entire Agreement; Corporate Power Governing Law Assignability	. 50 . 51 . 51 . 52 . 52 . 52 . 52 . 52 . 53 . 53
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1 10.2 10.3 10.4	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS Counterparts; Entire Agreement; Corporate Power Governing Law Assignability Third-Party Beneficiaries.	. 50 . 51 . 52 . 52 . 52 . 52 . 53 . 53 . 53 . 53
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1 10.2 10.3 10.4 10.5	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS Counterparts; Entire Agreement; Corporate Power Governing Law Assignability Third-Party Beneficiaries Notices	. 50 . 51 . 52 . 52 . 52 . 52 . 53 . 53 . 53 . 53 . 54
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1 10.2 10.3 10.4 10.5 10.6	Further Assurances Performance Order of Precedence E IX TERMINATION Termination E X MISCELLANEOUS Counterparts; Entire Agreement; Corporate Power Governing Law Assignability Third-Party Beneficiaries Notices Severability	. 50 . 51 . 52 . 52 . 52 . 52 . 52 . 53 . 53 . 53 . 53 . 54 . 54
8.1 8.2 8.3 ARTICL 9.1 ARTICL 10.1 10.2 10.3 10.4 10.5 10.6 10.7	Further Assurances. Performance Order of Precedence. E IX TERMINATION. Termination. E X MISCELLANEOUS	. 50 . 51 . 52 . 52 . 52 . 52 . 52 . 53 . 53 . 53 . 53 . 54 . 54 . 54

10.11	Headings	. 55
10.12	Survival of Covenants	. 55
10.13	Waivers of Default	. 55
10.14	Specific Performance	. 55
10.15	Amendments	. 55
10.16	Interpretation	. 56
10.17	Mutual Drafting	. 56
10.18	Limitations of Liability	. 56

ANNEX AND SCHEDULES

Annex A	Restructuring Steps Memorandum
Schedule 1.1A	Assumed Actions
Schedule 1.1B	Corner Store Business Exclusions
Schedule 1.1C	Corner Store Contracts
Schedule 1.1D	Corner Store Intellectual Property
Schedule 1.1E	Corner Store Software
Schedule 2.2(a)(i)	Members of the Corner Store Group following the Distribution
Schedule 2.2(a)(v)	Corner Store Assets
Schedule 2.2(b)(v)	Excluded Assets
Schedule 2.3(a)(ii)	Corner Store Liabilities
Schedule 2.3(b)(i)	Excluded Liabilities
Schedule 2.8(b)(ii)	Agreements between Corner Store Group and Valero Group that will not be terminated
Schedule 2.8(b)(iv)	Intercompany accounts payable or accounts receivable that will not be terminated
Schedule 2.9(a)	Shared Contracts
Schedule 5.5(g)	Special provisions applicable to certain pending Third-Party Claims
Schedule 5.9(a)	Guarantees, letters of credit and other obligations from which the Valero Group is to be removed
Schedule 6.1(c)	Insurance policies for which waivers of subrogation are required

SEPARATION AND DISTRIBUTION AGREEMENT

This SEPARATION AND DISTRIBUTION AGREEMENT, made and entered into effective as of April 29, 2013 (this "<u>Agreement</u>"), is by and between Valero Energy Corporation, a Delaware corporation ("<u>Valero</u>"), and CST Brands, Inc., a Delaware corporation and wholly owned subsidiary of Valero ("<u>Corner Store</u>"). Capitalized terms used herein and not otherwise defined shall have the respective meanings assigned to them in Article I.

RECITALS

WHEREAS, the Board of Directors of Valero (the "<u>Valero Board</u>") has determined that it is in the best interests of Valero and its stockholders to create a new publicly traded company that shall operate the Corner Store Business;

WHEREAS, Corner Store has been incorporated for this purpose and has not engaged in activities except in preparation for its corporate reorganization (including activities with respect to the Corner Store Financing Arrangements) and the distribution of its stock;

WHEREAS, in furtherance of the foregoing, the Valero Board has determined that it is appropriate and desirable for Valero and its applicable Subsidiaries to transfer the Corner Store Assets to Corner Store and certain entities designated by Corner Store that will be Subsidiaries of Corner Store as of the Distribution Date (any such entities, the "<u>Corner Store Designees</u>"), and for Corner Store and the Corner Store Designees to assume the Corner Store Liabilities, in each case as more fully described in this Agreement and the Ancillary Agreements (the "<u>Separation</u>");

WHEREAS, on the terms and conditions set forth in this Agreement, Valero shall contribute the Corner Store Assets to Corner Store and the Corner Store Designees in exchange for Corner Store Common Stock, the Senior Notes, the Cash Consideration and the assumption by Corner Store of the Corner Store Liabilities (the "<u>Contribution</u>");

WHEREAS, Valero currently intends that, on the Distribution Date, Valero shall distribute to holders of shares of Valero Common Stock, through a spin-off, 80% of the outstanding shares of Corner Store Common Stock, as more fully described in this Agreement and the Ancillary Agreements (the "<u>Distribution</u>");

WHEREAS, for U.S. federal income tax purposes, the Contribution and the Distribution, if effected, taken together, are intended to qualify as a tax-free transaction under Sections 355 and 368(a)(1)(D) of the Code;

WHEREAS, this Agreement is intended to be, and is hereby adopted as, a "plan of reorganization" within the meaning of Treas. Reg. Section 1.368-2(g); and

WHEREAS, it is appropriate and desirable to set forth the principal corporate transactions required to effect the Separation and the Distribution and certain other agreements that will govern certain matters relating to the Separation and the Distribution and the relationship of Valero, Corner Store and their respective Subsidiaries, following the Distribution.

NOW, THEREFORE, in consideration of the mutual agreements, provisions and covenants contained in this Agreement, the parties, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

For the purpose of this Agreement, the following terms shall have the following meanings:

"AAA Commercial Arbitration Rules" shall have the meaning set forth in Section 4.3(a).

"<u>Action</u>" means any demand, action, claim, dispute, suit, countersuit, arbitration, inquiry, subpoena, proceeding or investigation of any nature (whether criminal, civil, legislative, administrative, regulatory, prosecutorial or otherwise) by or before any federal, state, local, foreign or international Governmental Authority or any arbitration or mediation tribunal.

"<u>Affiliate</u>" means, when used with respect to a specified Person, a Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such specified Person. For the purpose of this definition, "<u>control</u>" (including with correlative meanings, "<u>controlled by</u>" and "<u>under common control with</u>"), when used with respect to any specified Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or other interests, by contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment, undertaking or otherwise. It is expressly agreed that, from and after the Distribution Date and for purposes of this Agreement and the Ancillary Agreements, no member of the Corner Store Group shall be deemed to be an Affiliate of any member of the Valero Group, and no member of the Valero Group shall be deemed to be an Affiliate of any member of the Valero Group.

"<u>Agent</u>" means the distribution agent to be appointed by Valero to distribute to the stockholders of Valero 80% of the outstanding shares of Corner Store Common Stock pursuant to the Distribution.

"Agreement" shall have the meaning set forth in the Preamble.

"<u>Ancillary Agreements</u>" means the Employee Matters Agreement, the Transition Services Agreements, the Tax Matters Agreement, the Registration Rights Agreement, the U.S. Fuel Supply Agreements, the Petroleum Product Supply Agreement, the Office Lease Agreement, the Office Sublease Agreement and the Transfer Documents.

"<u>Approvals or Notifications</u>" means any consents, waivers, approvals, permits or authorizations to be obtained from, notices, registrations or reports to be submitted to, or other filings to be made with, any third Person, including any Governmental Authority.

"<u>Assets</u>" means, with respect to any Person, the assets, properties, claims and rights (including goodwill) of such Person, wherever located (including in the possession of vendors or other third Persons or elsewhere), of every kind, character and description, whether real, personal or mixed, tangible, intangible or contingent, in each case, whether or not recorded or reflected or required

to be recorded or reflected on the books and records or financial statements of such Person, including the following:

(a) all accounting and other books, records and files whether in paper, microfilm, microfiche, computer tape or disc, magnetic tape, electronic or any other form;

(b) all apparatus, computers and other electronic data processing and communications equipment, fixtures, machinery, equipment, furniture, office equipment, automobiles, trucks, vessels, motor vehicles and other transportation equipment and other tangible personal property;

(c) all inventories of materials, parts, raw materials, components, supplies, works-in-process and finished goods and products;

(d) all interests in real property of whatever nature, including easements, whether as owner, mortgagee or holder of a Security Interest in real property, lessor, sublessor, lessee, sublessee or otherwise;

(e) (i) all interests in any capital stock or other equity interests of any Subsidiary, Affiliate or any other Person, (ii) all bonds, notes, debentures or other securities issued by any Subsidiary, Affiliate or any other Person, (iii) all loans, advances or other extensions of credit or capital contributions to any Subsidiary, Affiliate or any other Person and (iv) all other investments in securities of any Person;

(f) all license agreements, leases of personal property, open purchase orders for raw materials, supplies, parts or services and other contracts, agreements or commitments;

(g) all letters of credit;

(h) all written (including in electronic form) or oral technical information, data, specifications, research and development information, engineering drawings and specifications, operating and maintenance manuals, and materials and analyses prepared by consultants and other third Persons;

(i) all Intellectual Property and technology;

(j) all Software;

(k) all cost information, sales and pricing data, customer prospect lists, supplier records, customer and supplier lists, customer and vendor data, correspondence and lists, product data and literature, artwork, design, formulations and specifications, quality records and reports and other books, records, studies, surveys, reports, plans and documents;

(1) all prepaid expenses, trade accounts and other accounts and notes receivable;

(m) all rights under contracts or agreements, all claims or rights against any Person arising from the ownership of any Asset, all rights in connection with any bids or offers and all claims, choses in action or similar rights, whether accrued or contingent;

(n) all licenses, permits, approvals and authorizations which have been issued by any Governmental Authority;

(o) all cash or cash equivalents, bank accounts, lock boxes, security deposits and other deposit arrangements; and

(p) all interest rate, currency, commodity or other swap, collar, cap or other hedging or similar agreements or arrangements.

"Assumed Actions" means those Actions which are listed in Schedule 1.1A.

"<u>Cash Consideration</u>" means \$500,000,000 in cash to be paid to Valero by Corner Store in the Contribution.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Contribution</u>" shall have the meaning set forth in the Recitals.

"Corner Store" shall have the meaning set forth in the Preamble.

"Corner Store Accounts" shall have the meaning set forth in Section 2.10(a).

"Corner Store Assets" shall have the meaning set forth in Section 2.2(a).

"<u>Corner Store Balance Sheet</u>" means the audited combined balance sheet of the Corner Store Group, including the notes thereto, as of December 31, 2012.

"<u>Corner Store Business</u>" means (a) the retail business in the U.S. and Canada of Valero and its Subsidiaries and Affiliates (including, for the purposes of this definition, Corner Store and its Subsidiaries and Affiliates) prior to the date hereof as described in the Information Statement as the "retail business" and (b) without limiting the foregoing clause (a) and except as otherwise expressly provided in this Agreement, any terminated, divested or discontinued businesses, Assets or operations that were of such a nature that they would be part of the Corner Store Business (as described in the foregoing clause (a)) had they not been terminated, divested or discontinued (regardless of whether they ever operated under the "Corner Store" or "Dépanneur du Coin" name); <u>provided</u>, that the "Corner Store Business" shall not include any of the businesses, Assets or operations described on <u>Schedule 1.1B</u>.

"Corner Store Certificate of Incorporation" shall have the meaning set forth in Section 3.1(f).

"<u>Corner Store Common Stock</u>" means the common stock, par value \$0.01 per share, of Corner Store.

"Corner Store Contracts" means:

(a) any contract, agreement, arrangement, commitment or understanding with respect to which the only Persons party thereto or with rights, benefits or obligations thereunder or whose Assets are bound thereby are members of the Corner Store Group and third parties unaffiliated with the Valero Group;

(b) any employment, change of control, retention, consulting, indemnification, termination, severance or other similar agreements with any Corner Store Employee or consultants of the Corner Store Group;

(c) the contracts, agreements, arrangements, commitments or understandings listed on Schedule 1.1C; and

(d) any contract or agreement that is otherwise expressly contemplated pursuant to this Agreement or any of the Ancillary Agreements to be assigned to Corner Store or any member of the Corner Store Group.

"Corner Store Designees" shall have the meaning set forth in the Recitals.

"<u>Corner Store Employee</u>" means any individual who, immediately prior to the Distribution, is either actively employed primarily by, or then on an approved leave of absence from, any Person that will be a member of the Corner Store Group immediately after the Distribution.

"<u>Corner Store Financing Arrangements</u>" means the Senior Notes, the Revolving Credit Facility and the Term Loan Facility.

"<u>Corner Store Group</u>" means Corner Store, each Subsidiary of Corner Store immediately after the Distribution Date and each Affiliate of Corner Store immediately after the Distribution Date.

"Corner Store Indemnitees" shall have the meaning set forth in Section 5.3.

"Corner Store Intellectual Property" means the Intellectual property set forth on Schedule 1.1D.

"Corner Store Liabilities" shall have the meaning set forth in Section 2.3(a).

"Corner Store Software" means the Software set forth on Schedule 1.1E.

"Corner Store Transfer Documents" shall have the meaning set forth in Section 2.4(b).

"<u>Corporate Action</u>" means any Action, whether filed before, on or after the Distribution Date, to the extent it asserts violations of any federal, state, local, foreign or international securities Law, securities class action or shareholder derivative claim.

"<u>Credit Rating</u>" means on any date, the rating that has been most recently announced by any Rating Agency for any class of senior, unsecured, non-convertible publicly held long-term debt of a Person.

"<u>Custodial Party</u>" means the party that maintains the Records Facility where Stored Records are held.

"Dispute" shall have the meaning set forth in Section 4.1(a).

"Distribution" shall have the meaning set forth in the Recitals.

"Distribution Date" means the date and time at which the Distribution occurs.

"<u>Employee Matters Agreement</u>" means the Employee Matters Agreement, dated as of the date hereof, between Valero and Corner Store.

"<u>Environmental Law</u>" means any Law relating to pollution, protection or restoration of or prevention of harm to the environment or natural resources, including the use, handling, transportation, treatment, storage, disposal, Release or discharge of Hazardous Materials or the protection of or prevention of harm to human health and safety.

"<u>Environmental Liabilities</u>" means all Liabilities relating to, arising out of or resulting from any Hazardous Materials, Environmental Law or contract or agreement relating to environmental, health or safety matters (including all removal, remediation or cleanup costs, investigatory costs, response costs, natural resources damages, property damages, personal injury damages, costs of compliance, including with any product take-back requirements, or with any settlement, judgment or other determination of Liability and indemnity, contribution or similar obligations) and all costs and expenses, interest, fines, penalties or other monetary sanctions in connection therewith.

"<u>Exchange Act</u>" means the U.S. Securities Exchange Act of 1934, as amended, together with the rules and regulations promulgated thereunder.

"Excluded Assets" shall have the meaning set forth in Section 2.2(b).

"Excluded Liabilities" shall have the meaning set forth in Section 2.3(b).

"<u>Force Majeure</u>" means, with respect to a party, an event beyond the control of such party (or any Person acting on its behalf), which by its nature could not reasonably have been foreseen by such party (or such Person) or, if it could reasonably have been foreseen, was unavoidable, and includes acts of God, storms, floods, riots, fires, sabotage, civil commotion or civil unrest, interference by civil or military authorities, acts of war (declared or undeclared) or armed hostilities, other national or international calamities or acts of terrorism or failures of energy sources or distribution or transportation facilities. Notwithstanding the foregoing, the receipt by a party of an unsolicited takeover offer or other acquisition proposal, even if unforeseen or unavoidable, and such party's response thereto, shall not be deemed an event of Force Majeure.

"Form 10" shall have the meaning set forth in Section 3.3(a)(vii).

"<u>Governmental Approvals</u>" means any notices, reports or other filings to be made, or any consents, registrations, approvals, permits or authorizations to be obtained from, any Governmental Authority.

"<u>Governmental Authority</u>" means any nation or government, any state, municipality or other political subdivision thereof, and any entity, body, agency, commission, department, board,

bureau, court, tribunal or other instrumentality, whether federal, state, local, domestic, foreign or multinational, exercising executive, legislative, judicial, regulatory, administrative or other similar functions of, or pertaining to, government and any executive official thereof.

"Group" means either the Corner Store Group or the Valero Group, as the context requires.

"<u>Hazardous Materials</u>" means any chemical, material, substance, waste, pollutant, emission, discharge, release or contaminant that could result in liability under, or that is prohibited, limited or regulated by or pursuant to, any Environmental Law, and any natural or artificial substance (whether solid, liquid or gas, noise, ion, vapor or electromagnetic) that could cause harm to human health or the environment, including petroleum, petroleum products and byproducts, asbestos and asbestos-containing materials, urea formaldehyde foam insulation, electronic, medical or infectious wastes, polychlorinated biphenyls, radon gas, radioactive substances, chlorofluorocarbons and all other ozone-depleting substances.

"Indemnifying Party" shall have the meaning set forth in Section 5.4(a).

"Indemnitee" shall have the meaning set forth in Section 5.4(a).

"Indemnity Payment" shall have the meaning set forth in Section 5.4(a).

"<u>Information</u>" means information, whether or not patentable or copyrightable, in written, oral, electronic or other tangible or intangible forms, stored in any medium, including studies, reports, records, books, contracts, instruments, surveys, discoveries, ideas, concepts, know-how, techniques, designs, specifications, drawings, blueprints, diagrams, models, prototypes, samples, flow charts, data, computer data, disks, diskettes, tapes, computer programs or other software, marketing plans, customer names, memos, and other technical, financial, employee or business information or data.

"Information Statement" shall have the meaning set forth in Section 3.3(a)(vii).

"Initial Notice" shall have the meaning set forth in Section 4.2.

"Insurance Proceeds" means those monies:

- (a) received by an insured from an insurance carrier; or
- (b) paid by an insurance carrier on behalf of the insured;

in any such case net of any applicable premium adjustments (including reserves and retrospectively rated premium adjustments) and net of any costs or expenses incurred in the collection thereof; <u>provided</u>, <u>however</u>, that with respect to a captive insurance arrangement, Insurance Proceeds shall only include net amounts received by the captive insurer from a Third Party in respect of any captive reinsurance arrangement.

"<u>Intellectual Property</u>" means all of the following whether arising under the Laws of the United States or of any other foreign or multinational jurisdiction: (a) patents, patent applications (including patents issued thereon) and statutory invention registrations, including reissues, divisions, continuations, continuations in part, substitutions, renewals, extensions and reexaminations of any of the foregoing, and all rights in any of the foregoing provided by international treaties or conventions, (b) trademarks, service marks, trade names, service names, trade dress, logos and other source or business identifiers, including all goodwill associated with any of the foregoing and any and all common law rights in and to any of the foregoing, registrations and applications for registration of any of the foregoing, all rights in and to any of the foregoing provided by international treaties or conventions, and all reissues, extensions and renewals of any of the foregoing, (c) Internet domain names, (d) copyrightable works, copyrights, moral rights, mask work rights, database rights and design rights, whether or not registered, and all registrations and applications for registrations for registration of any of the foregoing, and all rights in and to any of the foregoing, and all registrations and applications for registration of any of the foregoing, and all rights in and to any of the foregoing, in the foregoing provided by international treaties or conventions, and all registrations and applications for registration of any of the foregoing, and all rights in and to any of the foregoing provided by international treaties or conventions, (e) confidential and proprietary information, including trade secrets, invention disclosures, processes and know-how and (f) intellectual property rights arising from or in respect of any technology.

"<u>Law</u>" means any national, supranational, federal, state, provincial, local or similar law (including common law), statute, code, order, ordinance, rule, regulation, treaty (including any income tax treaty), license, permit, authorization, approval, consent, decree, injunction, binding judicial or administrative interpretation or other requirement, in each case, enacted, promulgated, issued or entered by a Governmental Authority.

"Liabilities" means any and all debts, guarantees, assurances, commitments, liabilities, responsibilities, Losses, remediation, deficiencies, reimbursement obligations in respect of letters of credit, damages, fines, penalties, settlements, sanctions, costs, expenses, interest and obligations, whether accrued or fixed, absolute or contingent, matured or unmatured, accrued or not accrued, asserted or unasserted, liquidated or unliquidated, foreseen or unforeseen, known or unknown, reserved or unreserved, or determined or determinable, including those arising under any Law, claim (including any Third-Party Claim), demand, Action, or order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority or arbitration tribunal, and those arising under any contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment or undertaking, or any fines, damages or equitable relief that is imposed, in each case, including all costs and expenses relating thereto.

"<u>Losses</u>" means actual losses (including any diminution in value), costs, damages, fines, penalties and expenses (including legal and accounting fees and expenses and costs of investigation and litigation), whether or not involving a Third-Party Claim.

"Medicare Reporting Obligations" shall have the meaning set forth in Section 5.6(f).

"<u>Non-Custodial Party</u>" means the party that owns Stored Records held in the other party's Records Facility.

"<u>NYSE</u>" means the New York Stock Exchange.

"<u>Office Lease Agreement</u>" means the office lease agreement between a member of the Valero Group and a member of the Corner Store Group to be entered into in connection with the Separation.

"<u>Office Sublease Agreement</u>" means the office sublease agreement between a member of the Valero Group and a member of the Corner Store Group to be entered into in connection with the Separation.

"<u>Person</u>" means an individual, a general or limited partnership, a corporation, a trust, a joint venture, an unincorporated organization, a limited liability entity, any other entity and any Governmental Authority.

"<u>Petroleum Product Supply Agreement</u>" means the Petroleum Product Supply Agreement to be entered into between Valero Marketing and Supply Company and a member of the Corner Store Group in connection with the Separation.

"<u>Prime Rate</u>" means the rate which JPMorgan Chase Bank (or any successor thereto or other major money center commercial bank agreed to by the parties hereto) announces from time to time as its prime lending rate, as in effect from time to time.

"<u>Privileged Information</u>" means any information, in written, oral, electronic or other tangible or intangible forms, including any communications by or to attorneys (including attorney-client privileged communications), memoranda and other materials prepared by attorneys or under their direction (including attorney work product), as to which a party or its respective Subsidiaries would be entitled to assert or have a privilege, including the attorney-client and attorney work product privileges.

"<u>Rating Agency</u>" means Moody's Investors Service, Inc., Standard & Poor's, a division of The McGraw-Hill Companies, Inc., Fitch, Inc., any of their successors, or any nationally recognized statistical rating organizations registered with the SEC.

"<u>Record Date</u>" means 5:00 p.m. Eastern Time on the date to be determined by the Valero Board as the record date for determining stockholders of Valero entitled to receive shares of Corner Store Common Stock in the Distribution.

"Record Holders" means the holders of record of Valero Common Stock as of the Record Date.

"<u>Records Facility</u>" shall have the meaning set forth in Section 7.4(a).

"<u>Registration Rights Agreement</u>" means the Stockholder's and Registration Rights Agreement, dated as of April 29, 2013, between Valero and Corner Store.

"<u>Release</u>" means any release, spill, emission, discharge, leaking, pumping, pouring, dumping, injection, deposit, disposal, dispersal, leaching or migration of Hazardous Materials into the environment (including ambient air, surface water, groundwater and surface or subsurface strata).

"<u>Representatives</u>" means, with respect to any Person, any of such Person's directors, officers, employees, agents, consultants, advisors, accountants, attorneys or other representatives.

"<u>Response</u>" shall have the meaning set forth in Section 4.2.

"<u>Restructuring Steps Memorandum</u>" means the memorandum attached as <u>Annex A</u> setting forth the restructuring steps to be taken prior to the Distribution Date and the sequence thereof.

"<u>Retained Stock</u>" means the Corner Store Common Stock retained by Valero immediately following the Distribution.

"<u>Revolving Credit Facility</u>" means a revolving credit facility pursuant to a revolving credit facility agreement entered into prior to the Distribution Date by Corner Store, as borrower, the lender named therein as administrative agent, and the lenders named therein, on such terms and conditions as agreed to by Corner Store and the other parties to the revolving credit facility agreement and approved by Valero.

"SEC" means the U.S. Securities and Exchange Commission.

"<u>Security Interest</u>" means any mortgage, security interest, pledge, lien, charge, claim, option, right to acquire, voting or other restriction, right-of-way, covenant, condition, easement, encroachment, restriction on transfer, or other encumbrance of any nature whatsoever.

"<u>Senior Notes</u>" means the securities issued prior to the Distribution Date by Corner Store, and approved by Valero, in reliance on Rule 144A promulgated under the Securities Act.

"Separation" shall have the meaning set forth in the Recitals.

"Shared Contract" shall have the meaning set forth in Section 2.9(a).

"<u>Software</u>" means any and all (a) computer programs, including any and all software implementation of algorithms, models and methodologies, whether in source code, object code, human readable form or other form, (b) databases and compilations, including any and all data and collections of data, whether machine-readable or otherwise, (c) descriptions, flow charts and other work products used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons and (d) documentation, including user manuals and other training documentation, relating to any of the foregoing.

"<u>Stored Records</u>" means Tangible Information held in a Records Facility maintained or arranged for by the party other than the party that owns such Tangible Information.

"<u>Subsidiary</u>" or "<u>subsidiary</u>" means, with respect to any Person, any corporation, limited liability company, joint venture or partnership of which such Person (a) beneficially owns, either directly or indirectly, more than fifty percent (50%) of (i) the total combined voting power of all classes of voting securities of such Person, (ii) the total combined equity interests or (iii) the capital or profit interests, in the case of a partnership, or (b) otherwise has the power to vote, either directly or indirectly, sufficient securities to elect a majority of the board of directors or similar governing body.

"<u>Tangible Information</u>" means Information that is contained in written, electronic or other tangible forms.

"<u>Tax Matters Agreement</u>" means the Tax Matters Agreement, dated as of the date hereof, between Valero and Corner Store.

"Tax Return" shall have the meaning set forth in the Tax Matters Agreement.

"Taxes" shall have the meaning set forth in the Tax Matters Agreement.

"<u>Term Loan Facility</u>" means the term loan facility pursuant to the term loan agreement entered into prior to the Distribution Date by Corner Store, as borrower, the lender named therein as administrative agent, and the lenders named therein, on such terms and conditions as agreed to by Corner Store and the other parties to the term loan agreement and approved by Valero.

"<u>Third Party</u>" shall have the meaning set forth in Section 5.5(a).

"Third-Party Claim" shall have the meaning set forth in Section 5.5(a).

"Transfer Documents" shall have the meaning set forth in Section 2.4(b).

"Transferred Entity" shall have the meaning set forth in Section 2.2(a)(i).

"<u>Transition Services Agreements</u>" means the Transition Services Agreement, dated as of April 29, 2013, between Valero Services, Inc. and Corner Store, and the Transition Services Agreement, dated as of April 29, 2013, between Ultramar Ltd. and CST Canada Co.

"<u>U.S. Fuel Supply Agreements</u>" means a Branded Distributor Marketing Agreement, a Petroleum Product Sale Agreement and a Master Agreement to be entered into between a member of the Valero Group and a member of the Corner Store Group in connection with the Separation.

"Unreleased Corner Store Liability" shall have the meaning set forth in Section 2.6(b).

"Unreleased Excluded Liability" shall have the meaning set forth in Section 2.7(b).

"Valero" shall have the meaning set forth in the Preamble.

"Valero Accounts" shall have the meaning set forth in Section 2.10(a).

"Valero Board" shall have the meaning set forth in the Recitals.

"<u>Valero Business</u>" means each and every business conducted at any time by any member of the Valero Group, except the Corner Store Business.

"<u>Valero Common Stock</u>" means the common stock, par value \$0.01 per share, of Valero.

"<u>Valero Group</u>" means Valero, each Subsidiary of Valero immediately after the Distribution Date and each Affiliate of Valero immediately after the Distribution Date (in each case other than any member of the Corner Store Group).

"Valero Indemnitees" shall have the meaning set forth in Section 5.2.

"<u>Valero Intellectual Property</u>" means (a) the Valero Name and Valero Marks, and (b) all other Intellectual Property that, as of the Distribution Date, is owned or licensed by any member of either Group, other than the Corner Store Intellectual Property.

"<u>Valero Name and Valero Marks</u>" means the names, marks, trade dress, logos, monograms, domain names and other source or business identifiers of Valero or any of its Affiliates using or containing "Valero" or "Ultramar" (in block letters or otherwise), "Valero" or "Ultramar" either alone or in combination with other words or elements, and all names, marks, trade dress, logos, monograms, domain names and other source or business identifiers confusingly similar to or embodying any of the foregoing either alone or in combination with other words or elements, together with the goodwill associated with any of the foregoing.

"<u>Valero Software</u>" means all Software that, as of the Distribution Date, is owned or licensed by any member of either Group, other than the Corner Store Software.

"Valero Transfer Documents" shall have the meaning set forth in Section 2.1(b).

ARTICLE II THE SEPARATION

2.1 Transfer of Assets and Assumption of Liabilities.

(a) Unless otherwise provided in this Agreement or in any Ancillary Agreement, on or prior to the Distribution Date in accordance with the Restructuring Steps Memorandum and to the extent not previously effected prior to the date hereof pursuant to the steps of the Restructuring Steps Memorandum:

(i) Valero shall, and shall cause its applicable Subsidiaries to, assign, transfer, convey and deliver to Corner Store, or the applicable Corner Store Designees, and Corner Store or such Corner Store Designees shall accept from Valero and its applicable Subsidiaries, all of Valero's and such Subsidiaries' respective direct or indirect right, title and interest in and to all of the Corner Store Assets (it being understood that if any Corner Store Asset shall be held by a Transferred Entity or a wholly owned Subsidiary of a Transferred Entity, such Corner Store Asset may be assigned, transferred, conveyed and delivered as a result of the transfer of all or substantially all of the equity interests in such Transferred Entity);

(ii) Corner Store and the applicable Corner Store Designees shall accept, assume and agree faithfully to perform, discharge and fulfill all the Corner Store Liabilities in accordance with their respective terms. Corner Store and such Corner Store Designees shall be responsible for all Corner Store Liabilities, regardless of when or where such Corner Store Liabilities arose or arise, or whether the facts on which they are based occurred prior to or subsequent to the Distribution Date, regardless of where or against whom such Corner Store Liabilities are asserted or determined (including any Corner Store Liabilities arising out of claims made by the respective directors, officers, employees, agents, stockholders, Subsidiaries or Affiliates of either Group against any member of either Group) or whether asserted or determined prior to the date hereof, and regardless of whether arising from or alleged to arise from negligence, recklessness, violation of Law, fraud, misrepresentation or any other cause by any member of either Group, or any of their respective directors, officers, employees or agents;

(iii) Valero shall cause its applicable Subsidiaries to assign, transfer, convey and deliver to certain of its other Subsidiaries, which shall accept, such applicable Subsidiaries' respective right, title and interest in and to any Excluded Assets specified by Valero to be so assigned, transferred, conveyed and delivered; and

(iv) Valero and certain of its Subsidiaries shall accept and assume from certain of its other Subsidiaries and agree faithfully to perform, discharge and fulfill the Excluded Liabilities of such other Subsidiaries in accordance with their respective terms, and Valero and its applicable Subsidiaries shall be responsible for all Excluded Liabilities, regardless of when or where such Excluded Liabilities arose or arise, or whether the facts on which they are based occurred prior to or subsequent to the Distribution Date, regardless of where or against whom such Excluded Liabilities are asserted or determined (including any such Excluded Liabilities arising out of claims made by the respective directors, officers, employees, agents, stockholders, Subsidiaries or Affiliates of either Group against any member of either Group) or whether asserted or determined prior to the date hereof, and regardless of whether arising from or alleged to arise from negligence, recklessness, violation of Law, fraud, misrepresentation or any cause by any member of either Group, or any of their respective directors, officers, employees or agents.

In furtherance of the assignment, transfer, conveyance and delivery of the (b) Corner Store Assets and the assumption of the Corner Store Liabilities in accordance with Sections 2.1(a)(i) and 2.1(a)(ii), on, before or as of the date that such Corner Store Assets are assigned, transferred, conveyed or delivered or such Corner Store Liabilities are assumed, (i) Valero shall execute and deliver, and shall cause its Subsidiaries to execute and deliver, such transfer agreements, bills of sale, deeds, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent necessary to evidence the transfer, conveyance and assignment of all of Valero's and its Subsidiaries' (other than Corner Store and its Subsidiaries) right, title and interest in and to the Corner Store Assets to Corner Store and the Corner Store Designees, and (ii) Corner Store shall execute and deliver, and shall cause the Corner Store Designees to execute and deliver, such assumptions of contracts and other instruments of assumption as and to the extent necessary to evidence the valid and effective assumption of the Corner Store Liabilities by Corner Store and the Corner Store Designees. All of the documents contemplated by this Section 2.1(b) are referred to collectively herein as the "Valero Transfer Documents."

(c) To the extent that any Corner Store Asset is not transferred or assigned to, or any Corner Store Liability is not assumed by, a member of the Corner Store Group at the Distribution Date or is owned or held by a member of the Valero Group after the Distribution Date, from and after the Distribution Date, any such Corner Store Asset or Corner Store Liability shall be held by such member of the Valero Group for the use and benefit of the member of the Corner Store Group entitled thereto (at the expense of the member of the Corner Store Group entitled thereto in accordance with Section 2.5(c), and, subject to Section 2.5(b):

(i) Valero shall, and shall cause its applicable Subsidiaries to, as soon as reasonably practicable, assign, transfer, convey and deliver to Corner Store or certain of its Subsidiaries designated by Corner Store, and Corner Store or such Subsidiaries shall accept from Valero and its applicable Subsidiaries, all of Valero's and such Subsidiaries' respective right, title and interest in and to such Corner Store Assets; and

(ii) Corner Store and certain of its Subsidiaries designated by Corner Store shall, as soon as reasonably practicable, accept, assume and agree faithfully to perform, discharge and fulfill all such Corner Store Liabilities in accordance with their respective terms.

(d) Corner Store hereby waives compliance by each and every member of the Valero Group with the requirements and provisions of any "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer or sale of any or all of the Corner Store Assets to any member of the Corner Store Group.

(e) Valero hereby waives compliance by each and every member of the Corner Store Group with the requirements and provisions of any "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer or sale of any or all of the Excluded Assets to any member of the Valero Group.

2.2 Corner Store Assets.

(a) For purposes of this Agreement, "<u>Corner Store Assets</u>" means (without duplication):

(i) all issued and outstanding equity interests held by Valero or its Subsidiaries in the Subsidiaries of Valero that have been or shall be contributed to, or otherwise transferred, conveyed, or assigned to, the Corner Store Group or entities that shall be members of the Corner Store Group as of the Distribution Date, as listed on <u>Schedule 2.2(a)(i)</u> (such Subsidiaries and entities, the "<u>Transferred Entities</u>");

(ii) all Corner Store Contracts,

(iii) all Assets reflected as assets of Corner Store or its Subsidiaries on the Corner Store Balance Sheet, subject to any dispositions of such Assets subsequent to the date of the Corner Store Balance Sheet;

and

(iv) all Corner Store Intellectual Property and Corner Store Software;

(v) all Assets that are expressly provided by this Agreement or any Ancillary Agreement as Assets to be transferred to Corner Store or any other member of the Corner Store Group, including the Assets listed on <u>Schedule 2.2(a)(v)</u>.

Notwithstanding the foregoing, the Corner Store Assets shall not, in any event, include the Excluded Assets referred to in Section 2.2(b). All rights of the Corner Store Group in respect of

Valero insurance policies are set forth in Article V and shall not otherwise be included in the Corner Store Assets.

(b) For the purposes of this Agreement, "<u>Excluded Assets</u>" means (without duplication):

(i) all issued and outstanding equity interests held by Valero or its Subsidiaries in any entity that is not a member of the Corner Store Group;

(ii) any cash or cash equivalents withdrawn from Corner Store Accounts in accordance with Section 2.10(e);

(iii) the Valero Intellectual Property and Valero Software;

(iv) any Shared Contract;

(v) the Assets listed on <u>Schedule 2.2(b)(v)</u> and any and all other Assets that are expressly contemplated by this Agreement or any Ancillary Agreement as Assets to be retained by Valero or any other member of the Valero Group; and

(vi) any and all Assets of any members of the Valero Group that are not Corner Store Assets pursuant to Section 2.2(a).

2.3 Corner Store Liabilities.

(a) For the purposes of this Agreement, "<u>Corner Store Liabilities</u>" means (without duplication):

(i) all Liabilities, including any Environmental Liabilities, to the extent relating to, arising out of or resulting from:

(A) the operation or ownership of the Corner Store Business, as conducted at any time prior to, on or after the Distribution Date (including any Liability relating to, arising out of or resulting from any act or failure to act by any Person (whether or not such act or failure to act is or was within such Person's authority)) other than Liabilities arising under any Shared Contracts to the extent such Liabilities relate to the Valero Business; or

(B) any Corner Store Asset;

in any such case, whether arising before, on or after the Distribution Date;

(ii) the Liabilities listed on <u>Schedule 2.3(a)(ii)</u> and any and all other Liabilities that are expressly provided by this Agreement or any Ancillary Agreement as Liabilities to be assumed by Corner Store or any member of the Corner Store Group, and all agreements, obligations and Liabilities of any member of the Corner Store Group under this Agreement or any of the Ancillary Agreements; (iii) all Liabilities relating to, arising out of or resulting from the Corner Store Financing Arrangements;

(iv) all Liabilities reflected as liabilities or obligations of Corner Store or its Subsidiaries on the Corner Store Balance Sheet, subject to any discharge of such Liabilities subsequent to the date of the Corner Store Balance Sheet; and

(v) all Liabilities arising out of claims made by the respective directors, officers, stockholders, employees, agents, Subsidiaries or Affiliates of either Group against any member of either Group to the extent relating to, arising out of or resulting from the Corner Store Business or the other Liabilities referred to in clauses (i) through (iv) above, inclusive.

Notwithstanding the foregoing, the Corner Store Liabilities shall not include the Excluded Liabilities referred to in Section 2.3(b).

(b) For the purposes of this Agreement, "<u>Excluded Liabilities</u>" means (without duplication):

(i) the Liabilities listed on <u>Schedule 2.3(b)(i)</u> and any and all other Liabilities that are expressly contemplated by this Agreement or any Ancillary Agreement as Liabilities to be retained or assumed by Valero or any other member of the Valero Group;

(ii) all agreements and obligations of any member of the Valero Group under this Agreement or any of the Ancillary Agreements;

(iii) any and all Liabilities to the extent relating to, arising out of or resulting from any Excluded Assets (other than Liabilities arising under any Shared Contracts to the extent such Liabilities relate to the Corner Store Business), in any such case, whether arising before, on or after the Distribution Date;

(iv) all Liabilities arising out of claims made by the respective directors, officers, stockholders, employees, agents, Subsidiaries or Affiliates of either Group against any member of either Group to the extent relating to, arising out of or resulting from the Valero Business or the other Liabilities referred to in clauses (i) and (ii) above; and

(v) all Liabilities arising out of the ownership or operation, prior to the Distribution Date, by any entity that is a member of the Corner Store Group as of the Distribution Date, of (A) any petroleum refinery, truck rack, pipeline system or terminal; or (B) any other Asset never used in the Corner Store Business; or (C) with respect to other Assets that were, prior to the Distribution Date, owned by an entity that is a member of the Corner Store Group but used in both the Corner Store Business and other businesses or activities unrelated to the Corner Store Business ("<u>Other Businesses</u>"), that portion of any Liabilities arising from the use of those Assets in the Other Businesses (and for the purposes of this paragraph (v) the supply of petroleum products to the Corner Store Business shall constitute an Other Business).

2.4 <u>Transfer of Excluded Assets; Assumption of Excluded Liabilities.</u>

(a) To the extent that any Excluded Asset is transferred or assigned to, or any Excluded Liability is assumed by, a member of the Corner Store Group at the Distribution Date or is owned or held by a member of the Corner Store Group after the Distribution Date, from and after the Distribution Date, any such Excluded Asset or Excluded Liability shall be held by such member of the Corner Store Group for the use and benefit of the member of the Valero Group entitled thereto (at the expense of the member of the Valero Group entitled thereto) in accordance with Section 2.5(c), subject to Section 2.5(b) and:

(i) Corner Store shall, and shall cause its applicable Subsidiaries to, as soon as reasonably practicable, assign, transfer, convey and deliver to Valero or certain of its Subsidiaries designated by Valero, and Valero or such Subsidiaries shall accept from Corner Store and its applicable Subsidiaries, all of Corner Store's and such Subsidiaries' respective right, title and interest in and to such Excluded Assets; and

(ii) Valero and certain of its Subsidiaries designated by Valero shall, as soon as reasonably practicable, accept, assume and agree faithfully to perform, discharge and fulfill all such Excluded Liabilities in accordance with their respective terms.

(b) In furtherance of the assignment, transfer, conveyance and delivery of Excluded Assets and the assumption of Excluded Liabilities set forth in Sections 2.4(a)(i) and 2.4(a)(ii), and without any additional consideration therefor: (i) Corner Store shall execute and deliver, and shall cause its Subsidiaries to execute and deliver, such bills of sale, quitclaim deeds, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent necessary to evidence the transfer, conveyance and assignment of all of Corner Store's and its Subsidiaries' right, title and interest in and to the Excluded Assets to Valero and its Subsidiaries, and (ii) Valero shall execute and deliver, and shall cause its Subsidiaries to execute and deliver, such assumptions of contracts and other instruments of assumption as and to the extent necessary to evidence the valid and effective assumption of the Excluded Liabilities. All of the foregoing documents contemplated by this Section 2.4(b) and by Sections 2.1(a)(iii) and 2.1(a)(iv) shall be referred to collectively herein as the "Corner Store Transfer Documents" and, together with the Valero Transfer Documents, the "Transfer Documents."

2.5 Approvals and Notifications.

(a) To the extent that the transfer or assignment of any Corner Store Asset, the assumption of any Corner Store Liability, the transfer or assignment of any Excluded Asset, the assumption of any Excluded Liability, or the Separation or the Distribution requires any Approvals or Notifications, the parties shall endeavor to obtain or make such Approvals or Notifications as soon as reasonably practicable; <u>provided</u>, <u>however</u>, that, except to the extent expressly provided in this Agreement or any of the Ancillary Agreements or as otherwise agreed between Valero and Corner Store, neither Valero nor Corner Store shall be obligated to contribute capital or pay any consideration in any form (including providing any letter of credit,
guaranty or other financial accommodation) to any Person in order to obtain or make such Approvals or Notifications.

(b)If and to the extent that the valid, complete and perfected transfer or assignment to (i) the Corner Store Group of any Corner Store Assets or assumption by the Corner Store Group of any Corner Store Liabilities or (ii) the Valero Group of any Excluded Asset or assumption by the Valero Group of any Excluded Liability, would be a violation of applicable Law, or require any Approvals or Notifications in connection with the Separation or the Distribution that have not been obtained or made by the Distribution Date, then, unless the parties hereto shall otherwise mutually determine, the transfer or assignment to the Corner Store Group of such Corner Store Assets or to the Valero Group of such Excluded Assets, or the assumption by the Corner Store Group of such Corner Store Liabilities or the Valero Group of such Excluded Liabilities, as the case may be, shall be automatically deemed deferred and any such purported transfer, assignment or assumption shall be null and void until such time as all legal impediments are removed or such Approvals or Notifications have been obtained or made. Notwithstanding the foregoing, any such Corner Store Assets or Excluded Assets or Corner Store Liabilities or Excluded Liabilities shall continue to constitute Corner Store Assets, Excluded Assets, Corner Store Liabilities or Excluded Liabilities, as applicable, for all other purposes of this Agreement.

If any Corner Store Asset or Corner Store Liability is owned or held by (c) any member of the Valero Group after the Distribution Date, or any Excluded Asset or Excluded Liability is held by any member of the Corner Store Group after the Distribution Date, whether as a result of the provisions of Section 2.5(b) or for any other reason, then, insofar as reasonably possible, the member of the Corner Store Group holding or owning such Excluded Asset or such Excluded Liability, or the member of the Valero Group holding such Corner Store Asset or Corner Store Liability, as the case may be, shall thereafter hold such Corner Store Asset or Excluded Asset or Corner Store Liability or Excluded Liability, as the case may be, for the use and benefit of the member of the Valero Group or Corner Store Group entitled thereto (at the expense of the member of the Valero Group or Corner Store Group entitled thereto). In addition, the member of the Corner Store Group retaining such Excluded Asset or such Excluded Liability, or the member of the Valero Group retaining such Corner Store Asset or Corner Store Liability, shall, insofar as reasonably possible and to the extent permitted by applicable Law, treat such Corner Store Asset or Excluded Asset or Corner Store Liability or Excluded Liability in the ordinary course of business in accordance with past practice and take such other actions as may be reasonably requested by the member of the Corner Store Group or Valero Group to whom such Corner Store Asset or Excluded Asset is to be transferred or assigned, or which shall assume such Corner Store Liability or Excluded Liability, as the case may be, in order to place such member of the Corner Store Group or Valero Group in a substantially similar position as if such Corner Store Asset or Excluded Asset or Corner Store Liability or Excluded Liability had been transferred, assigned, assumed or retained in connection with the Separation to or by the party entitled thereto and so that all the benefits and burdens relating to such Corner Store Asset or Excluded Asset or Corner Store Liability or Excluded Liability, as the case may be, including use, risk of loss, potential for gain, and dominion, control and command over such Corner Store Asset or Excluded Asset or Corner Store Liability or Excluded Liability, as the case may be, and all costs and expenses related thereto, shall inure from and after the Distribution Date to the

Corner Store Group, in the case of any Corner Store Asset or Corner Store Liability, or the Valero Group, in the case of any Excluded Asset or Excluded Liability.

(d) If and when the Approvals or Notifications, the absence of which caused the deferral of transfer or assignment of any Corner Store Asset or Excluded Asset or the deferral of assumption of any Corner Store Liability or Excluded Liability pursuant to Section 2.5(b), are obtained or made, and, if and when any other legal impediments for the transfer or assignment of any Corner Store Asset or Excluded Asset or the assumption of any Corner Store Liability or Excluded Liability have been removed, the transfer or assignment of the applicable Corner Store Asset or Excluded Asset or the assumption of the applicable Corner Store Liability or Excluded Liability, as the case may be, shall be effected in accordance with the terms of this Agreement and/or the applicable Ancillary Agreement.

(e) Except as otherwise agreed between Valero and Corner Store, (i) any member of the Valero Group retaining a Corner Store Asset or Corner Store Liability (whether as a result of the provisions of Section 2.1(c), Section 2.5(b) or for any other reason) and (ii) any member of the Corner Store Group holding or owning an Excluded Asset or Excluded Liability (whether as a result of the provisions of Section 2.4(a), Section 2.5(b) or for any other reason), shall not be obligated, in order to effect the transfer of such Asset or Liability to the Group member entitled thereto, to expend any money unless the necessary funds are advanced (or otherwise made available) by the Group member entitled thereto, other than reasonable out-ofpocket expenses, attorneys' fees and recording or similar fees, all of which shall be promptly reimbursed by the Group member entitled to such Asset or Liability.

2.6 Novation of Corner Store Liabilities.

(a) Each of Valero and Corner Store, at the request of the other, shall endeavor, if reasonably practicable, to obtain, or to cause to be obtained, if reasonably practicable, any consent, substitution, approval or amendment required to novate or assign all obligations under agreements, leases, licenses and other obligations or Liabilities of any nature whatsoever that constitute Corner Store Liabilities, or to obtain in writing the unconditional release of all parties to such arrangements other than any member of the Corner Store Group, so that, in any such case, the members of the Corner Store Group shall be solely responsible for the Corner Store Liabilities; <u>provided</u>, <u>however</u>, that neither Valero nor Corner Store shall be obligated to contribute any capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any third Person from whom any such consent, substitution, approval, amendment or release is requested.

(b) If Valero or Corner Store is unable to obtain, or to cause to be obtained, any such required consent, substitution, approval, amendment or release, and the applicable member of the Valero Group continues to be bound by such agreement, lease, license or other obligation or Liability (each, an "<u>Unreleased Corner Store Liability</u>"), Corner Store shall, to the extent not prohibited by Law, as indemnitor, guarantor, agent or subcontractor for such member of the Valero Group, as the case may be, (i) pay, perform and discharge fully all the obligations or other Liabilities of such member of the Valero Group that constitute Unreleased Corner Store Liabilities from and after the Distribution Date and (ii) use its commercially reasonable efforts to effect such payment, performance or discharge prior to any demand for such payment, performance, or discharge is permitted to be made by the obligee thereunder on any member of the Valero Group. If and when any such consent, substitution, approval, amendment or release shall be obtained or the Unreleased Corner Store Liabilities shall otherwise become assignable or able to be novated, Valero shall promptly assign, or cause to be assigned, and Corner Store or the applicable Corner Store Group member shall assume, such Unreleased Corner Store Liabilities without exchange of further consideration.

2.7 Novation of Excluded Liabilities.

(a) Each of Valero and Corner Store, at the request of the other, shall endeavor, if reasonably practicable, to obtain, or to cause to be obtained, if reasonably practicable, any consent, substitution, approval or amendment required to novate or assign all obligations under agreements, leases, licenses and other obligations or Liabilities of any nature whatsoever that constitute Excluded Liabilities, or to obtain in writing the unconditional release of all parties to such arrangements other than any member of the Valero Group, so that, in any such case, the members of the Valero Group shall be solely responsible for such Excluded Liabilities; <u>provided</u>, <u>however</u>, that neither Valero nor Corner Store shall be obligated to contribute any capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any third Person from whom any such consent, substitution, approval, amendment or release is requested.

If Valero or Corner Store is unable to obtain, or to cause to be obtained, (b)any such required consent, substitution, approval, amendment or release and the applicable member of the Corner Store Group continues to be bound by such agreement, lease, license or other obligation or Liability (each, an "Unreleased Excluded Liability"), Valero shall, to the extent not prohibited by Law, as indemnitor, guarantor, agent or subcontractor for such member of the Corner Store Group, as the case may be, (i) pay, perform and discharge fully all the obligations or other Liabilities of such member of the Corner Store Group that constitute Unreleased Excluded Liabilities from and after the Distribution Date and (ii) use its commercially reasonable efforts to effect such payment, performance, or discharge prior to any demand for such payment, performance, or discharge is permitted to be made by the obligee thereunder on any member of the Corner Store Group. If and when any such consent, substitution, approval, amendment or release shall be obtained or the Unreleased Excluded Liabilities shall otherwise become assignable or able to be novated, Corner Store shall promptly assign, or cause to be assigned, and Valero or the applicable Valero Group member shall assume, such Unreleased Excluded Liabilities without exchange of further consideration.

2.8 <u>Termination of Agreements</u>.

(a) Except as set forth in Section 2.8(b), in furtherance of the releases and other provisions set forth in Article V, Corner Store and each member of the Corner Store Group, on the one hand, and Valero and each member of the Valero Group, on the other hand, hereby terminate any and all agreements, arrangements, commitments or understandings, whether or not in writing, between or among Corner Store and/or any member of the Corner Store Group and/or any entity that shall be a member of the Corner Store Group as of the Distribution Date, on the one hand, and Valero and/or any member of the Valero Group (other than entities that shall be members of the Corner Store Group as of the Distribution Date, on the one hand, and Valero and/or any member of the Distribution Date), on the other hand, effective as

of the Distribution Date. No such terminated agreement, arrangement, commitment or understanding (including any provision thereof which purports to survive termination) shall be of any further force or effect after the Distribution Date. Each party shall, at the reasonable request of any other party, take, or cause to be taken, such other actions as may be necessary to effect the foregoing.

(b) The provisions of Section 2.8(a) shall not apply to any of the following agreements, arrangements, commitments or understandings (or to any of the provisions thereof): (i) this Agreement and the Ancillary Agreements (and each other agreement or instrument expressly contemplated by this Agreement or any Ancillary Agreement to be entered into by any of the parties hereto or any of the members of their respective Groups, including, for the avoidance of doubt, those agreements and instruments entered into in connection with the Corner Store Financing Arrangements); (ii) any agreements, arrangements, commitments or understandings filed as an exhibit, whether in preliminary or final form, to the Form 10 or otherwise listed or described on Schedule 2.8(b)(ii); (iii) any agreements, arrangements, commitments or understandings to which any Person other than the parties hereto and the members of their respective Groups is a party (it being understood that to the extent that the rights and obligations of the parties and the members of their respective Groups under any such agreements, arrangements, commitments or understandings constitute Corner Store Assets or Corner Store Liabilities, they shall be assigned pursuant to Section 2.1); (iv) any intercompany accounts payable or accounts receivable described on Schedule 2.8(b)(iv); (v) any agreements, arrangements, commitments or understandings to which any member of the Valero Group or Corner Store Group, other than Valero, Corner Store or any wholly owned Subsidiary of Valero or Corner Store, as the case may be, is a party (it being understood that directors' qualifying shares or similar interests shall be disregarded for purposes of determining whether a Subsidiary is wholly owned); (vi) any Shared Contracts; and (vii) any other agreements, arrangements, commitments or understandings that this Agreement or any Ancillary Agreement expressly contemplates shall survive the Distribution Date.

2.9 Treatment of Shared Contracts.

Without limiting the generality of the obligations set forth in Section 2.1, (a) unless the parties otherwise agree or the benefits of any contract, agreement, arrangement, commitment or understanding described in this Section 2.9 are expressly conveyed to the applicable party pursuant to an Ancillary Agreement, (i) any contract, agreement, arrangement, commitment or understanding that is listed on Schedule 2.9(a) shall be assigned in part to the applicable member(s) of the applicable Group, if so assignable, or appropriately amended prior to, on or after the Distribution Date, so that each party or the members of its respective Group shall, as of the Distribution Date, be entitled to the rights and benefits, and shall assume the related portion of any Liabilities, inuring to its respective businesses, in each case, in accordance with the allocation of benefits and burdens set forth on Schedule 2.9(a), and (ii) (A) any contract, agreement, arrangement, commitment or understanding that is an Excluded Asset or Excluded Liability but, prior to the Distribution Date, inured in part to the benefit or burden of any member of the Corner Store Group, and (B) any contract, agreement, arrangement, commitment or understanding that is a Corner Store Asset or a Corner Store Liability but, prior to the Distribution Date, inured in part to the benefit or burden of any member of the Valero Group, shall be assigned in part to the applicable member(s) of the applicable Group, if so assignable, or

appropriately amended prior to, on or after the Distribution Date, so that each party or the members of its respective Group shall, as of the Distribution Date, be entitled to the rights and benefits, and shall assume the related portion of any Liabilities, inuring to its respective businesses (any contract, agreement, arrangement, commitment or understanding referred to in clause (i) or (ii) above, a "Shared Contract"); provided, however, that, in the case of each of clause (i) and (ii), (1) in no event shall any member of any Group be required to assign (or amend) any Shared Contract in its entirety or to assign a portion of any Shared Contract which is not assignable (or cannot be amended) by its terms (including any terms imposing consents or conditions on an assignment where such consents or conditions have not been obtained or fulfilled), and (2) if any Shared Contract cannot be so partially assigned by its terms or otherwise, or cannot be amended or if such assignment or amendment would impair the benefit the parties thereto derive from such Shared Contract, then the parties shall, and shall cause each of their respective Subsidiaries to, take such other reasonable and permissible actions (including by providing prompt notice to the other party with respect to any relevant claim of Liability or other relevant matters arising in connection with a Shared Contract so as to allow such other party the ability to exercise any applicable rights under such Shared Contract) to cause a member of the Corner Store Group or the Valero Group, as the case may be, to receive the rights and benefits of that portion of each Shared Contract that relates to the Corner Store Business or the businesses retained by Valero, as the case may be (in each case, to the extent so related), as if such Shared Contract had been assigned to (or amended to allow) a member of the applicable Group pursuant to this Section 2.9, and to bear the burden of the corresponding Liabilities (including any Liabilities that may arise by reason of such arrangement), as if such Liabilities had been assumed by a member of the applicable Group pursuant to this Section 2.9.

(b) Each of Valero and Corner Store shall, and shall cause the members of its Group to, (i) treat for all Tax purposes the portion of each Shared Contract inuring to its respective businesses as Assets owned by, and/or Liabilities of, as applicable, such party, or its subsidiaries, as applicable, not later than the Distribution Date, and (ii) neither report nor take any Tax position (on a Tax Return or otherwise) inconsistent with such treatment (unless required by applicable Law).

(c) Nothing in this Section 2.9 shall require any member of any Group to make any material payment (except to the extent advanced, assumed or agreed in advance to be reimbursed by any member of the other Group), incur any material obligation or grant any material concession for the benefit of any member of any other Group in order to effect any transaction contemplated by this Section 2.9.

2.10 Bank Accounts; Cash Balances.

(a) Valero and Corner Store each agrees to take, or cause the respective members of their respective Groups to take, at the Distribution Date (or such earlier time as Valero and Corner Store may agree), all actions necessary to amend all contracts or agreements governing each bank and brokerage account owned by Corner Store or any other member of the Corner Store Group (collectively, the "<u>Corner Store Accounts</u>") so that such Corner Store Accounts, if currently linked (whether by automatic withdrawal, automatic deposit or any other authorization to transfer funds from or to, hereinafter "linked") to any bank or brokerage account

owned by Valero or any other member of the Valero Group (collectively, the "<u>Valero</u> <u>Accounts</u>"), are de-linked from the Valero Accounts.

(b) Valero and Corner Store each agrees to take, or cause the respective members of their respective Groups to take, at the Distribution Date (or such earlier time as Valero and Corner Store may agree), all actions necessary to amend all agreements governing the Valero Accounts so that such Valero Accounts, if currently linked to a Corner Store Account, are de-linked from the Corner Store Accounts.

(c) It is intended that, following consummation of the actions contemplated by Sections 2.10(a) and 2.10(b), there shall be in place a centralized cash management process pursuant to which the Corner Store Accounts shall be managed centrally and funds collected shall be transferred into one or more centralized accounts maintained by Corner Store.

(d) It is intended that, following consummation of the actions contemplated by Sections 2.10(a) and 2.10(b), there shall continue to be in place a centralized cash management process pursuant to which the Valero Accounts shall be managed centrally and funds collected shall be transferred into one or more centralized accounts maintained by Valero.

(e) With respect to any outstanding payments initiated by Valero, Corner Store, or any of their respective Subsidiaries prior to the Separation, such outstanding payments shall be honored following the Separation by the Person or Group owning the account from which the payment was initiated.

(f) As between Valero and Corner Store (and the members of their respective Groups), all payments made and reimbursements received after the Separation by either party (or member of its Group) that relate to a business, Asset or Liability of the other party (or member of its Group), shall be held by such party for the use and benefit of the party entitled thereto (at the expense of the party entitled thereto). Each party shall maintain an accounting of any such payments and reimbursements, and the parties shall have a monthly reconciliation, whereby all such payments made and reimbursements received by each party are calculated and the net amount owed to Valero or Corner Store shall be paid over with right of set-off. If at any time the net amount owed to either party entitled thereto within three (3) business days of such amount exceeding \$10,000,000. Notwithstanding the foregoing, neither Valero nor Corner Store shall act as collection agent for the other party, nor shall either party act as surety or endorser with respect to non-sufficient funds checks, or funds to be returned in a bankruptcy or fraudulent conveyance action.

2.11 <u>Other Ancillary Agreements</u>. Effective as of the date hereof, each of Valero and Corner Store shall execute and deliver all Ancillary Agreements to which it is a party (other than the Transfer Documents, which shall be executed on or prior to the Distribution Date).

2.12 <u>Disclaimer of Representations and Warranties</u>. EACH OF VALERO (ON BEHALF OF ITSELF AND EACH MEMBER OF THE VALERO GROUP) AND CORNER STORE (ON BEHALF OF ITSELF AND EACH MEMBER OF THE CORNER STORE GROUP) UNDERSTANDS AND AGREES THAT, EXCEPT AS EXPRESSLY SET FORTH

HEREIN OR IN ANY ANCILLARY AGREEMENT, NO PARTY TO THIS AGREEMENT. ANY ANCILLARY AGREEMENT OR ANY OTHER AGREEMENT OR DOCUMENT CONTEMPLATED BY THIS AGREEMENT, ANY ANCILLARY AGREEMENT OR OTHERWISE, IS REPRESENTING OR WARRANTING IN ANY WAY AS TO THE ASSETS, BUSINESSES OR LIABILITIES TRANSFERRED OR ASSUMED AS CONTEMPLATED HEREBY OR THEREBY, AS TO ANY CONSENTS OR APPROVALS **REQUIRED IN CONNECTION THEREWITH, AS TO THE VALUE OR FREEDOM FROM** ANY SECURITY INTERESTS OF, OR ANY OTHER MATTER CONCERNING, ANY ASSETS OF SUCH PARTY, AS TO, IN THE CASE OF INTELLECTUAL PROPERTY. NON-INFRINGEMENT OR ANY WARRANTY THAT ANY SUCH INTELLECTUAL PROPERTY IS "ERROR FREE," OR AS TO THE ABSENCE OF ANY DEFENSES OR RIGHT OF SET-OFF OR FREEDOM FROM COUNTERCLAIM WITH RESPECT TO ANY CLAIM OR OTHER ASSET, INCLUDING ANY ACCOUNTS RECEIVABLE, OF ANY PARTY, OR AS TO THE LEGAL SUFFICIENCY OF ANY ASSIGNMENT, DOCUMENT OR INSTRUMENT DELIVERED HEREUNDER TO CONVEY TITLE TO ANY ASSET OR THING OF VALUE UPON THE EXECUTION, DELIVERY AND FILING HEREOF OR THEREOF. EXCEPT AS MAY EXPRESSLY BE SET FORTH HEREIN OR IN ANY ANCILLARY AGREEMENT, ALL SUCH ASSETS ARE BEING TRANSFERRED ON AN "AS IS," "WHERE IS" BASIS (AND, IN THE CASE OF ANY REAL PROPERTY, EXCEPT AS OTHERWISE AGREED BY VALERO, BY MEANS OF A QUITCLAIM OR SIMILAR FORM DEED OR CONVEYANCE) AND THE RESPECTIVE TRANSFEREES SHALL BEAR THE ECONOMIC AND LEGAL RISKS THAT (I) ANY CONVEYANCE WILL PROVE TO BE INSUFFICIENT TO VEST IN THE TRANSFEREE GOOD AND MARKETABLE TITLE, FREE AND CLEAR OF ANY SECURITY INTEREST, AND (II) ANY NECESSARY APPROVALS OR NOTIFICATIONS ARE NOT OBTAINED OR THAT ANY REQUIREMENTS OF LAWS OR JUDGMENTS ARE NOT COMPLIED WITH.

ARTICLE III

THE DISTRIBUTION

3.1 <u>Actions Prior to the Distribution</u>. Prior to the Distribution Date and subject to the terms and conditions set forth herein, the parties shall take, or cause to be taken, the following actions in connection with the Distribution:

(a) *Notice to NYSE*. Valero shall, to the extent possible, give the NYSE not less than ten (10) days' advance notice of the Record Date in compliance with Rule 10b-17 under the Exchange Act.

(b) Securities Law Matters. Corner Store shall file any amendments or supplements to the Form 10 as may be necessary or advisable in order to cause the Form 10 to become and remain effective as required by the SEC or federal, state or other applicable securities Laws. Valero and Corner Store shall cooperate in preparing, filing with the SEC and causing to become effective registration statements or amendments thereof which are required to reflect the establishment of, or amendments to, any employee benefit and other plans necessary or advisable in connection with the transactions contemplated by this Agreement and the Ancillary Agreements. Valero and Corner Store shall take all such action as may be necessary or advisable under the securities or blue sky Laws of the United States (and any comparable Laws under any non-U.S. jurisdiction) in connection with the transactions contemplated by this Agreement and the Ancillary Agreements.

(c) *Mailing of Information Statement*. Valero shall, as soon as is reasonably practicable after the Form 10 is declared effective under the Exchange Act and the Valero Board has approved the Distribution, cause the Information Statement to be mailed to the Record Holders.

(d) *The Distribution Agent*. Valero shall enter into a distribution agent agreement with the Agent or otherwise provide instructions to the Agent regarding the Distribution.

(e) Stock-Based Employee Benefit Plans. At or prior to the Distribution Date, Valero and Corner Store shall take all actions as may be necessary to approve the stock-based employee benefit plans of Corner Store in order to satisfy the requirements of Rule 16b-3 under the Exchange Act and the applicable rules and regulations of the NYSE.

(f) *Certificate of Incorporation.* Valero and Corner Store shall take all necessary action that may be required to provide for the adoption by Corner Store of the Amended and Restated Certificate of Incorporation of Corner Store (the "<u>Corner Store</u> <u>Certificate of Incorporation</u>") and the Amended and Restated Bylaws of Corner Store, each in such form as may be reasonably determined by Valero and Corner Store, and Corner Store shall file the Corner Store Certificate of Incorporation with the Secretary of State of the State of Delaware.

(g) Financing Arrangements. Corner Store shall enter into the Corner Store Financing Arrangements, on such terms and conditions as agreed by Valero (including the amount that shall be borrowed pursuant to the Corner Store Financing Arrangements and the interest rates for such borrowings). Valero and Corner Store shall participate in the preparation of all materials and presentations as may be reasonably necessary to secure funding pursuant to the Corner Store Financing Arrangements, including Rating Agency presentations necessary to obtain the requisite ratings needed to secure the financing under any of the Corner Store Financing Arrangements and any marketing efforts or road shows related thereto. The parties agree that Corner Store, and not Valero, shall be ultimately responsible for all third-party costs and expenses incurred by, and for reimbursement of such costs and expenses to, any member of the Valero Group or Corner Store Group associated with the Corner Store Financing Arrangements.

(h) *Restructuring Steps Memorandum*. Valero and Corner Store shall take the steps set forth on the Restructuring Steps Memorandum to the extent the Restructuring Steps Memorandum contemplates such steps being taken at or prior to the Distribution Date.

(i) Satisfying Conditions to Distribution. Valero and Corner Store shall cooperate to cause the conditions to the Distribution set forth in Section 3.3 to be satisfied and to effect the Distribution at the Distribution Date.

3.2 Effecting the Distribution.

(a) *Delivery of Corner Store Common Stock.* On or prior to the Distribution Date, Valero shall deliver to the Agent, for the benefit of the Record Holders, book-entry transfer authorizations for such number of the outstanding shares of Corner Store Common Stock as is necessary to effect the Distribution.

(b) *Distribution Date*. The Distribution Date shall be 11:59 p.m. Eastern Time, or such other time as Valero may determine, on such date as Valero may determine.

(c) *Distribution of Shares and Cash.* Valero shall instruct the Agent to distribute, as soon as practicable following the Distribution Date, to each Record Holder the following: (i) one share of Corner Store Common Stock for each nine shares of Valero Common Stock held by such Record Holder as of the Record Date and (ii) cash, if applicable, in lieu of fractional shares obtained in the manner provided in Section 3.2(d).

No Fractional Shares. No fractional shares shall be distributed or credited (d) to book-entry accounts in connection with the Distribution. As soon as practicable after the Distribution Date, Valero shall direct the Agent to determine the number of whole shares and fractional shares of Corner Store Common Stock allocable to each holder of record or beneficial owner of Valero Common Stock as of the Record Date, to aggregate all such fractional shares and to sell the whole shares obtained thereby in open market transactions (with the Agent, in its sole and absolute discretion, determining when, how and through which broker-dealer and at what price to make such sales), and to cause to be distributed to each such holder or for the benefit of each such beneficial owner, in lieu of any fractional share, such holder's or owner's ratable share of the proceeds of such sale, after deducting any taxes required to be withheld and after deducting an amount equal to all brokerage charges, commissions and transfer taxes attributed to such sale. Neither Valero nor Corner Store shall be required to guarantee any minimum sale price for the fractional shares of Corner Store Common Stock. Neither Valero nor Corner Store shall be required to pay any interest on the proceeds from the sale of fractional shares.

(e) *Beneficial Owners*. Solely for purposes of computing fractional share interests pursuant to Section 3.2(d), the beneficial owner of Valero Common Stock held of record in the name of a nominee in any nominee account shall be treated as the holder of record with respect to such shares.

(f) Unclaimed Stock or Cash. Any Corner Store Common Stock or cash in lieu of fractional shares with respect to Corner Store Common Stock that remains unclaimed by any Record Holder one hundred eighty (180) days after the Distribution Date shall be delivered to Corner Store. Corner Store shall hold such Corner Store Common Stock for the account of such Record Holder, and the parties agree that all obligations to provide such Corner Store Common Stock and cash, if any, in lieu of fractional share interests shall be obligations of Corner Store, subject in each case to applicable escheat or other abandoned property Laws, and Valero shall have no Liability with respect thereto. (g) *Transfer Authorizations*. Corner Store agrees to provide all book-entry transfer authorizations for shares of Corner Store Common Stock that Valero or the Agent shall require in order to effect the Distribution.

3.3 <u>Conditions to the Distribution</u>.

(a) The consummation of the Distribution shall be subject to the satisfaction, or waiver by Valero in its sole and absolute discretion, of the conditions set forth in this Section 3.3(a).

(i) *Completion of the Separation*. The Separation shall have been completed in accordance with the Restructuring Steps Memorandum.

(ii) *Private Letter Ruling.* Valero shall have received a private letter ruling from the U.S. Internal Revenue Service substantially to the effect that, among other things, the Contribution and the Distribution, if effected, taken together, shall qualify as a transaction that is tax-free for U.S. federal income tax purposes under Sections 355 and 368(a)(1)(D) of the Code.

(iii) Opinion of Nationally Recognized Accounting Firm. Valero shall have received an opinion from a nationally recognized accounting firm in form and substance satisfactory to Valero, in its sole discretion, to the effect that the Contribution and the Distribution, if effected, taken together, shall qualify as a transaction that is tax-free for U.S. federal income tax purposes under Sections 355 and 368(a)(1)(D) of the Code.

(iv) *Governmental Approvals*. All Governmental Approvals necessary to consummate the Distribution shall have been obtained and be in full force and effect.

(v) Securities Laws. The actions and filings necessary or appropriate under applicable securities Laws in connection with the Distribution shall have been taken or made, and, where applicable, have become effective or been accepted by the applicable Governmental Authority.

(vi) *No Order or Injunction.* No order, injunction or decree issued by any court or agency of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the Distribution or any of the related transactions shall be in effect, and no other event outside the control of Valero shall have occurred or failed to occur that prevents the consummation of the Distribution or any of the related transactions.

(vii) Effectiveness of Form 10; Mailing of Information Statement. A Registration Statement on Form 10 registering the Corner Store Common Stock (the "Form 10") shall be effective under the Exchange Act, with no stop order in effect with respect thereto, and the Information Statement included therein (the "Information Statement") shall have been mailed to Valero's stockholders as of the Record Date. (viii) *Listing on NYSE*. The Corner Store Common Stock shall have been accepted for listing on the NYSE, subject to official notice of distribution.

(ix) *Contribution.* Valero shall have received the Cash Consideration, the Senior Notes and Corner Store Common Stock in connection with the Contribution and shall be satisfied in its sole discretion that, as of the Distribution Date, it shall have no further Liability whatsoever under the Corner Store Financing Arrangements (including in connection with any guarantees provided by any member of the Valero Group).

(x) *Execution of Ancillary Agreements*. Each of the Ancillary Agreements shall have been duly executed and delivered by the parties thereto.

(xi) Solvency Opinion. Valero shall have received an opinion from a nationally recognized investment banking firm or other authority in form and substance satisfactory to Valero, in its sole discretion, confirming the solvency and financial viability of Corner Store after consummation of the Distribution, and such opinion shall not have been withdrawn or rescinded.

(xii) No Circumstances Making Distribution Inadvisable. No events or developments shall have occurred or exist that, in the judgment of the Valero Board, in its sole and absolute discretion, make it inadvisable to effect the Distribution or the other transactions contemplated hereby, or would result in the Distribution or the other transactions contemplated hereby not being in the best interest of Valero or its stockholders.

3.4 <u>Sole Discretion</u>. The foregoing conditions are for the sole benefit of Valero and shall not give rise to or create any duty on the part of Valero or the Valero Board to waive or not waive such conditions or in any way limit Valero's right to terminate this Agreement as set forth in Article IX or alter the consequences of any such termination from those specified in such Article. Any determination made by the Valero Board prior to the Distribution concerning the satisfaction or waiver of any or all of the conditions set forth in Section 3.3 shall be conclusive.

ARTICLE IV DISPUTE RESOLUTION

4.1 <u>General Provisions</u>.

(a) Any dispute, controversy or claim arising out of or relating to this Agreement or the Ancillary Agreements (except as otherwise set forth in any such Ancillary Agreements), including the validity, interpretation, breach or termination thereof (a "<u>Dispute</u>"), shall be resolved in accordance with the procedures set forth in this Article IV, which shall be the sole and exclusive procedures for the resolution of any such Dispute unless otherwise specified in the applicable Ancillary Agreement or in this Article IV.

(b) THE PARTIES EXPRESSLY WAIVE AND FOREGO ANY RIGHT TO TRIAL BY JURY.

(c) The specific procedures set forth in this Article IV, including the time limits referenced herein, may be modified by agreement of both of the parties in writing.

(d) All applicable statutes of limitations and defenses based upon the passage of time shall be tolled while the procedures specified in this Article IV are pending. The parties shall take any necessary or appropriate action required to effectuate such tolling.

4.2 <u>Consideration by Senior Executives</u>. If a Dispute is not resolved in the normal course of business at the operational level, the parties shall attempt in good faith to resolve the Dispute by negotiation between executives who hold, at a minimum, the office of Senior Vice President and/or General Counsel. Either party may initiate the executive negotiation process by providing a written notice to the other (the "<u>Initial Notice</u>"). Within fifteen (15) days after delivery of the Initial Notice, the receiving party shall submit to the other a written response (the "<u>Response</u>"). The Initial Notice and the Response shall include (a) a statement of the Dispute and of each party's position and (b) the names and titles of the executives who may represent that party and of any other persons who may accompany the executive. The parties agree that such executives shall have full and complete authority to resolve any Disputes submitted pursuant to this Section 4.2. Such executives shall meet in person or by teleconference or video conference within thirty (30) days of the date of the Initial Notice to seek a resolution of the Dispute. In the event that the executives are unable to agree to a format for such meeting, the meeting shall be convened in person at a mutually acceptable location in San Antonio, Texas.

4.3 <u>Arbitration</u>.

(a) In the event any Dispute is not finally resolved pursuant to Section 4.2 within sixty (60) days from the delivery of the Initial Notice, and unless the parties have mutually agreed to mediate or use some other form of alternative dispute resolution in an attempt to resolve the Dispute, then such Dispute may be submitted to be finally resolved by binding arbitration pursuant to the AAA Commercial Arbitration Rules as then in effect (the "<u>AAA</u> <u>Commercial Arbitration Rules</u>").

(b) Without waiving its rights to any remedy under this Agreement and without first complying with the provisions of Section 4.2, either party may seek any interim or provisional relief that is necessary to protect the rights or property of that party either (i) before any Texas federal or state court, (ii) before a special arbitrator, as provided for under the AAA Commercial Arbitration Rules, or (iii) before the arbitral tribunal established hereunder.

(c) Unless otherwise agreed by the parties in writing, any Dispute to be decided in arbitration hereunder shall be decided (i) before a sole arbitrator if the amount in dispute, inclusive of all claims and counterclaims, totals less than \$3 million; or (ii) by an arbitral tribunal of three (3) arbitrators if (A) the amount in dispute, inclusive of all claims and counterclaims, is equal to or greater than \$3 million or (B) either party elects in writing to have such dispute decided by three (3) arbitrators when one of the parties believes, in its sole judgment, the issue could have significant precedential value; however, the party who makes that request shall solely bear the increased costs and expenses associated with a panel of three (3) arbitrators).

(d) The panel of three (3) arbitrators shall be chosen as follows: (i) upon the written demand of either party and within fifteen (15) days from the date of receipt of such demand, each party shall name an arbitrator selected by such party in its sole discretion; and (ii) the two (2) party-appointed arbitrators shall thereafter, within thirty (30) days from the date on which the second of the two (2) arbitrators was named, name a third, independent arbitrator who shall act as chairperson of the arbitral tribunal. In the event that either party fails to name an arbitrator within fifteen (15) days from the date of receipt of a written demand to do so, then upon written application by either party, that arbitrator shall be appointed pursuant to the AAA Commercial Arbitration Rules. In the event that the two (2) party-appointed arbitrators fail to appoint the third, independent arbitrator within thirty (30) days from the date on which the second of the two (2) arbitrators was named, then upon written application by either party, the third, independent arbitrator shall be appointed pursuant to AAA Commercial Arbitration Rules. If the arbitration shall be before a sole independent arbitrator, then the sole independent arbitrator shall be appointed by agreement of the parties within fifteen (15) days from the date of receipt of written demand of either party. If the parties cannot agree to a sole independent arbitrator, then upon written application by either party, the sole independent arbitrator shall be appointed pursuant to AAA Commercial Arbitration Rules.

(e) The place of arbitration shall be San Antonio, Texas. Along with the arbitrator(s) appointed, the parties shall agree to a mutually convenient date and time to conduct the arbitration, but in no event shall the final hearing(s) be scheduled more than nine (9) months from submission of the Dispute to arbitration unless the parties agree otherwise in writing.

(f) The arbitral tribunal shall have the right to award, on an interim basis, or include in the final award, any relief that it deems proper in the circumstances, including money damages (with interest on unpaid amounts from the due date), injunctive relief (including specific performance) and attorneys' fees and costs; <u>provided</u> that the arbitral tribunal shall not award any relief not specifically requested by the parties and, in any event, shall not award any damages of the types prohibited under Section 10.18. Upon constitution of the arbitral tribunal following any grant of interim relief by a special arbitrator or court pursuant to Section 4.3(b), the tribunal may affirm or disaffirm that relief, and the parties shall seek modification or rescission of the order entered by the special arbitrator or court as necessary to accord with the tribunal's decision.

(g) Neither party shall be bound by Rule 13 of the Federal Rules of Civil Procedure or any analogous Law or provision in the AAA Commercial Arbitration Rules governing deadlines for compulsory counterclaims; rather, each party shall be free to bring a counterclaim at any time (subject to any applicable statutes of limitation).

(h) So long as either party has a timely claim to assert, the agreement to arbitrate Disputes set forth in this Section 4.3 shall continue in full force and effect subsequent to, and notwithstanding the completion, expiration or termination of, this Agreement.

(i) The interim or final award in an arbitration pursuant to this Article IV shall be conclusive and binding upon the parties, and a party obtaining a final award may enter judgment upon such award in any court of competent jurisdiction.

(j) It is the intent of the parties that the agreement to arbitrate Disputes set forth in this Section 4.3 shall be interpreted and applied broadly such that all reasonable doubts as to arbitrability of a Dispute shall be decided in favor of arbitration.

(k) The parties agree that any Dispute submitted to arbitration shall be governed by, and construed and interpreted in accordance with Texas Law, as provided in Section 10.2 and, except as otherwise provided in this Article IV or mutually agreed to in writing by the parties, the Federal Arbitration Act, 9 U.S.C. §§ 1 *et seq.*, shall govern any arbitration between the parties pursuant to this Section 4.3.

(1) Subject to Section 4.3(c)(ii)(B), each party shall bear its own fees, costs and expenses and shall bear an equal share of the costs and expenses of the arbitration, including the fees, costs and expenses of the three (3) arbitrators; <u>provided</u> that the arbitral tribunal may award the prevailing party its reasonable fees and expenses (including attorneys' fees), if it finds that there was no good faith basis for the position taken by the other party in the arbitration.

(m) Notwithstanding anything in this Article IV to the contrary, any disputes relating to the interpretation of Article V or requesting injunctive relief or specific performance shall be conducted according to the fast-track arbitration procedures of the AAA Commercial Arbitration Rules then in effect.

ARTICLE V MUTUAL RELEASES; INDEMNIFICATION

5.1 <u>Release of Pre-Distribution Claims</u>.

Except as provided in Section 5.1(c), effective as of the Distribution Date. (a) Corner Store does hereby, for itself and each other member of the Corner Store Group, their respective Affiliates (other than any member of the Valero Group), successors and assigns, and all Persons who at any time prior to the Distribution Date have been directors, officers, agents or employees of any member of the Corner Store Group (in each case, in their respective capacities as such), remise, release and forever discharge Valero and the members of the Valero Group, their respective Affiliates (other than any member of the Corner Store Group), successors and assigns, and all Persons who at any time prior to the Distribution Date have been stockholders, directors, officers, agents or employees of any member of the Valero Group (in each case, in their respective capacities as such), and their respective heirs, executors, administrators, successors and assigns, from any and all Liabilities whatsoever, whether at Law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of Law or otherwise, existing or arising from any acts or events occurring or failing to occur or alleged to have occurred or to have failed to occur or any conditions existing or alleged to have existed on or before the Distribution Date, including in connection with the transactions related to or undertaken in connection with the Separation and the Distribution and all other activities to implement the Separation and the Distribution or contemplated hereunder.

(b) Except as provided in Section 5.1(c), effective as of the Distribution Date, Valero does hereby, for itself and each other member of the Valero Group, their respective Affiliates (other than any member of the Corner Store Group), successors and assigns, and all Persons who at any time prior to the Distribution Date have been directors, officers, agents or employees of any member of the Valero Group (in each case, in their respective capacities as such), remise, release and forever discharge Corner Store, the respective members of the Corner Store Group, their respective Affiliates (other than any member of the Valero Group), successors and assigns, and all Persons who at any time prior to the Distribution Date have been stockholders, directors, officers, agents or employees of any member of the Corner Store Group (in each case, in their respective capacities as such), and their respective heirs, executors, administrators, successors and assigns, from any and all Liabilities whatsoever, whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, existing or arising from any acts or events occurring or failing to occur or alleged to have occurred or to have failed to occur or any conditions existing or alleged to have existed on or before the Distribution Date, including in connection with the transactions related to or undertaken in connection with the Separation and the Distribution and all other activities to implement the Separation and the Distribution or contemplated hereunder.

(c) Nothing contained in Section 5.1(a) or (b) shall impair any right of any Person to enforce this Agreement, any Ancillary Agreement or any agreements, arrangements, commitments or understandings that are specified in Section 2.8(b) or the applicable Schedules thereto as not to terminate as of the Distribution Date, in each case in accordance with its terms. Nothing contained in Section 5.1(a) or (b) shall release any Person from:

(i) any Liability provided in or resulting from any agreement among any members of the Valero Group or the Corner Store Group that is specified in Section 2.8(b) or the applicable Schedules thereto as not to terminate as of the Distribution Date, or any other Liability specified in such Section 2.8(b) as not to terminate as of the Distribution Date;

(ii) any Liability, contingent or otherwise, assumed, transferred, assigned or allocated to the Group of which such Person is a member in accordance with, or any other Liability of any member of any Group under, this Agreement or any Ancillary Agreement (including any Corner Store Liability and any Excluded Liability, as applicable);

(iii) any Liability for the sale, lease, construction or receipt of goods, property or services purchased, obtained or used in the ordinary course of business by a member of one Group from a member of the other Group prior to the Distribution Date;

(iv) any Liability for unpaid amounts for products or services or refunds owing on products or services due on a value-received basis for work done by a member of one Group at the request or on behalf of a member of the other Group;

(v) any Liability that the parties may have with respect to indemnification or contribution pursuant to this Agreement for claims brought against the parties by third Persons, which Liability shall be governed by the provisions of this Article V and Article VI and any other applicable provisions of this Agreement or any Ancillary Agreement; or (vi) any Liability the release of which would result in the release of any third Person other than a Person released pursuant to this Section 5.1.

In addition, nothing contained in Section 5.1(a) shall release Valero from honoring its existing obligations to indemnify any director, officer or employee of a member of the Corner Store Group who was a director, officer or employee of a member of the Valero Group on or prior to the Distribution Date, to the extent that such director, officer or employee becomes a named defendant in any Action with respect to which such director, officer or employee was entitled to such indemnification pursuant to then-existing obligations, it being understood that, if the underlying obligation giving rise to such Action is a Corner Store Liability, Corner Store shall indemnify Valero for such Liability (including Valero's costs to indemnify the director, officer or employee) in accordance with the provisions set forth in this Article V.

(d) Corner Store covenants that it shall not make, and shall not permit any member of the Corner Store Group to make, any claim or demand, or commence any Action asserting any claim or demand, including any claim of contribution or any indemnification, against Valero or any member of the Valero Group, or any other Person released pursuant to Section 5.1(a), with respect to any Liabilities released pursuant to Section 5.1(a). Valero covenants that it shall not make, and shall not permit any member of the Valero Group to make, any claim or demand, or commence any Action asserting any claim or demand, including any claim of contribution or any indemnification, against Corner Store or any member of the Corner Store Group, or any other Person released pursuant to Section 5.1(b), with respect to any Liabilities released pursuant to Section 5.1(b), with respect to any Liabilities released pursuant to Section 5.1(b).

(e) It is the intent of each of Valero and Corner Store, by virtue of the provisions of this Section 5.1, to provide for a full and complete release and discharge of all Liabilities existing or arising from all acts and events occurring or failing to occur or alleged to have occurred or to have failed to occur and all conditions existing or alleged to have existed on or before the Distribution Date, between or among Corner Store or any member of the Corner Store Group, on the one hand, and Valero or any member of the Valero Group, on the other hand (including any contractual agreements or arrangements existing or alleged to exist between or among any such members on or before the Distribution Date), except as expressly set forth in Section 5.1(c). At any time, at the request of any other party to this Agreement, each party shall cause each member of its respective Group to execute and deliver releases reflecting the provisions hereof.

(f) Any breach of the provisions of this Section 5.1 by either Valero or Corner Store shall entitle the other party to recover reasonable fees and expenses of counsel in connection with such breach or any Action resulting from such breach.

5.2 <u>Indemnification by Corner Store</u>. Subject to Section 5.4, Corner Store shall, and shall cause the other members of the Corner Store Group to, indemnify, defend and hold harmless Valero, each member of the Valero Group and each of their respective directors, officers and employees, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "<u>Valero Indemnitees</u>"), from and against any and all Liabilities of the Valero Indemnitees relating to, arising out of or resulting from any of the following items (without duplication):

(a) the failure of Corner Store or any other member of the Corner Store Group or any other Person to pay, perform or otherwise promptly discharge any Corner Store Liabilities or Corner Store Contracts in accordance with their respective terms, whether prior to or after the Distribution Date or the date hereof;

(b) the Corner Store Business, any Corner Store Liabilities or any Corner Store Contracts;

(c) the Assumed Actions;

(d) any Corporate Action or Action relating exclusively to the Corner Store Business from which Corner Store is unable to cause a Valero Group party to be removed pursuant to Section 5.6(d);

(e) any failure by Corner Store or a member of the Corner Store Group to use commercially reasonable efforts to obtain the waivers of subrogation contemplated by Section 5.4(d);

(f) any breach by Corner Store or any member of the Corner Store Group of this Agreement or any of the Ancillary Agreements;

(g) any guarantee, indemnification obligation, letter of credit reimbursement obligations, surety, bond or other credit support agreement, arrangement, commitment or understanding for the benefit of Corner Store or its Subsidiaries by Valero or any of its Subsidiaries (other than Corner Store or its Subsidiaries) that survives following the Distribution Date; and

(h) any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, with respect to all information contained in any of the Form 10 (including in any amendments or supplements thereto), the Information Statement (as amended or supplemented if Corner Store shall have furnished any amendments or supplements thereto), any registration statement relating to the Retained Stock or the Senior Notes or any offering memorandum or other marketing materials prepared in connection with the Corner Store Financing Arrangements, other than any such statement or omission in the Form 10, Information Statement, such registration statements or offering memorandum or other marketing materials to the extent provided in writing by Valero and primarily concerning the Valero Group.

5.3 <u>Indemnification by Valero</u>. Subject to Section 5.4, Valero shall, and shall cause the other members of the Valero Group to, indemnify, defend and hold harmless Corner Store, each member of the Corner Store Group and each of their respective directors, officers and employees, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "<u>Corner Store Indemnitees</u>"), from and against any and all Liabilities of the Corner Store Indemnitees relating to, arising out of or resulting from any of the following items (without duplication): (a) the failure of Valero or any other member of the Valero Group or any other Person to pay, perform or otherwise promptly discharge any Excluded Liabilities, whether prior to or after the Distribution Date or the date hereof;

- (b) the Valero Business and the Excluded Assets;
- (c) the Excluded Liabilities;

(d) any Corporate Action or Action relating exclusively to the Valero Business from which Valero is unable to cause a Corner Store Group party to be removed pursuant to Section 5.6(d);

(e) any failure by Valero or a member of the Valero Group to use commercially reasonable efforts to obtain the waivers of subrogation contemplated by Section 5.4(d);

(f) any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, with respect to all information contained in any of the Form 10 (including in any amendments or supplements thereto), the Information Statement (as amended or supplemented if Corner Store shall have furnished any amendments or supplements thereto), any registration statement relating to the Retained Stock or the Senior Notes or any offering memorandum or other marketing materials prepared in connection with the Corner Store Financing Arrangements, in each case only to the extent provided in writing by Valero and solely concerning the Valero Group; and

(g) any breach by Valero or any member of the Valero Group of this Agreement or any Ancillary Agreements.

5.4 Indemnification Obligations Net of Insurance Proceeds.

(a) The parties intend that any Liability subject to indemnification or reimbursement pursuant to this Article V or Article VI shall be net of Insurance Proceeds that actually reduce the amount of the Liability. Accordingly, the amount that any party (an "<u>Indemnifying Party</u>") is required to pay to any Person entitled to indemnification hereunder (an "<u>Indemnitee</u>") shall be reduced by any Insurance Proceeds theretofore actually recovered by or on behalf of the Indemnitee in respect of the related Liability. If an Indemnifying Party in respect of any Liability and subsequently receives Insurance Proceeds, then the Indemnitee shall pay to the Indemnifying Party an amount equal to the excess of the Indemnity Payment received over the amount of the Indemnity Payment that would have been due if the Insurance Proceeds had been received, realized or recovered before the Indemnity Payment was made.

(b) An insurer who would otherwise be obligated to pay any claim shall not be relieved of the responsibility with respect thereto or, solely by virtue of the indemnification provisions hereof, have any subrogation rights with respect thereto, it being expressly understood and agreed that no insurer or any other Third Party shall be entitled to a "windfall" (*i.e.*, a benefit they would not be entitled to receive in the absence of the indemnification provisions hereof) by virtue of the indemnification provisions hereof.

(c) Each of Valero and Corner Store shall, and shall cause the members of its Group to, when appropriate, use commercially reasonable efforts to obtain waivers of subrogation for each of the insurance policies identified on <u>Schedule 6.1(c)</u>. Each of Valero and Corner Store hereby waives, for itself and each member of its Group, its rights to recover against the other party in subrogation or as subrogee for a third Person.

(d) For all claims as to which indemnification is provided under Section 5.2 or 5.3 other than Third-Party Claims (as to which Section 5.5 shall apply), the reasonable fees and expenses of counsel to the Indemnitee for the enforcement of the indemnity obligations shall be borne by the Indemnifying Party.

5.5 <u>Procedures for Indemnification of Third-Party Claims</u>.

(a) If an Indemnitee shall receive written notice from a Person (including any Governmental Authority) who is not a member of the Valero Group or the Corner Store Group (a "<u>Third Party</u>") of any claim or of the commencement by any such Person of any Action (collectively, a "<u>Third-Party Claim</u>") with respect to which an Indemnifying Party may be obligated to provide indemnification to such Indemnitee pursuant to Section 5.2 or 5.3, or any other Section of this Agreement or, subject to Section 5.14, any Ancillary Agreement, such Indemnitee shall give such Indemnifying Party written notice thereof within fourteen (14) days of receipt of such written notice. Any such notice shall describe the Third-Party Claim in reasonable detail and include copies of all notices and documents (including court papers) received by the Indemnitee to provide notice in accordance with this Section 5.5(a) shall not relieve an Indemnifying Party of its indemnification obligations under this Agreement, except to the extent to which the Indemnifying Party shall demonstrate that it was materially prejudiced by the Indemnitee's failure to provide notice in accordance with this Section 5.5(a).

(b) An Indemnifying Party may elect to defend (and to seek to settle or compromise), at such Indemnifying Party's own expense and by such Indemnifying Party's own counsel, any Third-Party Claim. Within thirty (30) days after the receipt of notice from an Indemnitee in accordance with Section 5.5(a) (or sooner, if the nature of such Third-Party Claim so requires), the Indemnifying Party shall notify the Indemnitee of its election whether the Indemnifying Party shall assume responsibility for defending such Third-Party Claim. After notice from an Indemnifying Party to an Indemnitee of its election to assume the defense of a Third-Party Claim, such Indemnitee shall have the right to employ separate counsel and to participate in (but not control) the defense, compromise, or settlement thereof, but the fees and expenses of such counsel shall be the expense of such Indemnitee except as otherwise expressly set forth herein.

(c) If an Indemnifying Party has elected to assume the defense of a Third-Party Claim, then such Indemnifying Party shall be solely liable for all fees and expenses incurred by it in connection with the defense of such Third-Party Claim and shall not be entitled to seek any indemnification or reimbursement from the Indemnitee for any such fees or expenses incurred during the course of its defense of such Third Party Claim, regardless of any subsequent decision by the Indemnifying Party to reject or otherwise abandon its assumption of such defense. If an Indemnifying Party elects not to assume responsibility for defending any Third-Party Claim or fails to notify an Indemnitee of its election within thirty (30) days after receipt of a notice from an Indemnitee, and the Indemnitee conducts and controls the defense of such Third-Party Claim, then the Indemnifying Party shall be liable for all reasonable fees and expenses incurred by the Indemnitee in connection with the defense of such Third-Party Claim.

(d) Notwithstanding an election by an Indemnifying Party to defend a Third-Party Claim pursuant to Section 5.5(b), a Indemnitee may, upon notice to the Indemnifying Party, elect to take over the defense of such Third-Party Claim if (i) in its exercise of reasonable business judgment, the Indemnitee determines that the Indemnifying Party is not defending such Third-Party Claim competently or in good faith, (ii) the Credit Rating of the Indemnifying Party as determined by at least two Rating Agencies is or falls below "B" as established by Standard & Poor's or Fitch, Inc., or "B2" as established by Moody's, or the equivalent as established by any other Rating Agency, (iii) the Indemnitee determines in its exercise of reasonable business judgment that there exists a compelling business reason for such Indemnitee to defend such Third-Party Claim (other than as contemplated by the foregoing clause (i)), (iv) the Indemnifying Party makes a general assignment for the benefit of creditors, has filed against it or files a petition in bankruptcy or insolvency or is declared bankrupt or insolvent or declares that it is bankrupt or insolvent, or (v) there occurs a change of control of the Indemnifying Party.

An Indemnitee that does not conduct and control the defense of any Third-(e) Party Claim, or an Indemnifying Party that has failed to elect to defend any Third-Party Claim as contemplated hereby, nevertheless shall have the right to employ separate counsel (including local counsel as necessary) of its own choosing to monitor and participate in (but not control) the defense of any Third-Party Claim for which it is a potential Indemnitee or Indemnifying Party, but the fees and expenses of such counsel shall be at the expense of such Indemnitee or Indemnifying Party, as the case may be, and the provisions of Section 5.5(c) shall not apply to such fees and expenses. Notwithstanding the foregoing, subject to Section 7.7, such party shall cooperate with the party entitled to conduct and control the defense of such Third-Party Claim in such defense and make available to the controlling party, at the non-controlling party's expense, all witnesses, information and materials in such party's possession or under such party's control relating thereto as are reasonably required by the controlling party. In addition to the foregoing, if any Indemnitee shall in good faith determine that such Indemnitee and the Indemnifying Party have actual or potential differing defenses or conflicts of interest between them that make joint representation inappropriate, then the Indemnitee shall have the right to employ separate counsel (including local counsel as necessary) and to participate in (but not control) the defense, compromise or settlement thereof, and the Indemnifying Party shall bear the reasonable fees and expenses of such counsel for all Indemnitees.

(f) Neither party may settle or compromise any Third-Party Claim for which either party is seeking to be indemnified hereunder without the prior written consent of the other party, which consent may not be unreasonably withheld, unless such settlement or compromise is solely for monetary damages, does not involve any finding or determination of wrongdoing or violation of Law by the other party and provides for a full, unconditional and irrevocable release of the other party from all Liability in connection with the Third-Party Claim. The parties hereby agree that if a party presents the other party with a written notice containing a proposal to settle or compromise a Third-Party Claim for which either party is seeking to be indemnified hereunder and the party receiving such Proposal does not respond in any manner to the party presenting such proposal within thirty (30) days (or within any such shorter time period that may be required by applicable Law or court order) of receipt of such proposal, then the party receiving such proposal shall be deemed to have consented to the terms of such proposal.

(g) Schedule 5.5(g) identifies certain pending Third-Party Claims with respect to which Liabilities will be allocated and the other actions taken as set forth therein. With respect to the Third-Party Claims identified in Schedule 5.5(g), in the event of any conflict between the provisions of this Article V and the provisions of Schedule 5.5(g), the latter shall govern. There shall be no requirement under this Section 5.5 to give notice with respect to any Third-Party Claims that exist as of the Distribution Date.

(h) The provisions of this Section 5.5 (other than this Section 5.5(h)) and the provisions of Section 5.6 shall not apply to Taxes (Taxes being governed by the Tax Matters Agreement).

(i) All Assumed Actions have been tendered by Valero to Corner Store and are deemed to be formally accepted by Corner Store upon the execution of this Agreement.

(j) The Indemnifying Party shall establish a procedure reasonably acceptable to the Indemnitee to keep the Indemnitee reasonably informed of the progress of the Third-Party Claim and to notify the Indemnitee when any such Third-Party Claim is closed, regardless of whether such Third-Party Claim was resolved by settlement, verdict, dismissal or otherwise.

5.6 Additional Matters.

(a) Indemnification payments in respect of any Liabilities for which an Indemnitee is entitled to indemnification under this Article V shall be paid by the Indemnifying Party to the Indemnitee as such Liabilities are incurred upon demand by the Indemnitee, including reasonably satisfactory documentation setting forth the basis for the amount of such indemnification payment, including documentation with respect to calculations made and consideration of any Insurance Proceeds that actually reduce the amount of such Liabilities. THE INDEMNITY AGREEMENTS CONTAINED IN THIS ARTICLE V SHALL REMAIN OPERATIVE AND IN FULL FORCE AND EFFECT, REGARDLESS OF (I) ANY INVESTIGATION MADE BY OR ON BEHALF OF ANY INDEMNITEE, (II) THE KNOWLEDGE BY THE INDEMNITEE OF LIABILITIES FOR WHICH IT MIGHT BE ENTITLED TO INDEMNIFICATION HEREUNDER AND (III) ANY TERMINATION OF THIS AGREEMENT.

(b) Any claim on account of a Liability that does not result from a Third-Party Claim shall be asserted by written notice given by the Indemnitee to the related Indemnifying Party. Such Indemnifying Party shall have a period of thirty (30) days after the receipt of such notice within which to respond thereto. If such Indemnifying Party does not respond within such thirty (30)-day period, such Indemnifying Party shall be deemed to have refused to accept responsibility to make payment. If such Indemnifying Party does not respond within such thirty (30)-day period or rejects such claim in whole or in part, such Indemnitee shall be free to pursue such remedies as may be available to such party as contemplated by this Agreement and the Ancillary Agreements.

(c) In the event of payment by or on behalf of any Indemnifying Party to any Indemnitee in connection with any Third-Party Claim, such Indemnifying Party shall be subrogated to and shall stand in the place of such Indemnitee as to any events or circumstances in respect of which such Indemnitee may have any right, defense or claim relating to such Third-Party Claim against any claimant or plaintiff asserting such Third-Party Claim or against any other Person. Such Indemnitee shall cooperate with such Indemnifying Party in a reasonable manner, and at the cost and expense of such Indemnifying Party, in prosecuting any subrogated right, defense or claim.

(d) In the event of an Action for which indemnification is sought pursuant to Section 5.2 or 5.3 and in which the Indemnifying Party is not a named defendant, if either the Indemnifying Party shall so request, the parties shall use commercially reasonable efforts to substitute the Indemnifying Party for the named defendant.

(e) In the event that Corner Store or Valero establishes a risk accrual in an amount of at least \$5 million with respect to any Third-Party Claim for which such party has indemnified the other party pursuant to Section 5.2 or 5.3, as applicable, it shall notify the other party of the existence and amount of such risk accrual (*i.e.*, when the accrual is recorded in the financial statements as an accrual for a potential liability), subject to the parties entering into an appropriate agreement with respect to the confidentiality and/or privilege thereof.

Each of Corner Store and Valero shall provide the other party or a Third (f)Party designated by such party with all information necessary for such party to comply with its obligations under Section 111 of the Medicare, Medicaid and SCHIP Extension Act (or any successor thereto) ("Medicare Reporting Obligations") with respect to the settlement or other disposition of any Action by or on behalf of such party. If Corner Store or Valero fails to do so promptly and/or provides materially incorrect information, then such party shall indemnify the other party pursuant for any fines, penalties and/or costs arising from any such failure or action. Each of Corner Store and Valero shall bear its own costs associated with satisfying such Medicare Reporting Obligations (including but not limited to settlements or releases of personal injury claims from a Medicare beneficiary on behalf of such party). Each of Corner Store and Valero shall provide the other party with a monthly written report identifying all Actions that are subject to Medicare Reporting Obligations on the part of any member of the applicable Group and that have been settled or otherwise disposed of by or on behalf of any member of such Group. In addition, Corner Store and Valero may establish a procedure reasonably acceptable to each of Corner Store and Valero to automatically send electronic notice to the other party through the litigation management system or any successor system when any such Action is closed, regardless of whether such Action was decided by settlement, verdict, dismissal or was otherwise disposed of.

(g) An Indemnitee shall take all reasonable steps to mitigate damages in respect of any claim for which it seeks indemnification hereunder, and shall use reasonable efforts to avoid any costs or expenses associated with such claim and, if such costs and expenses

cannot be avoided, to minimize the amount thereof; <u>provided</u>, <u>however</u>, that an Indemnitee shall have no obligation to make a claim for recovery against any of its insurers with respect to any Losses for which it is seeking indemnification.

(h) THE RELEASES AND INDEMNIFICATION OBLIGATIONS OF THE PARTIES IN THIS AGREEMENT ARE EXPRESSLY INTENDED, AND SHALL OPERATE AND BE CONSTRUED, TO APPLY EVEN WHERE THE LOSSES OR LIABILITIES FOR WHICH THE RELEASE AND/OR INDEMNITY ARE GIVEN ARE CAUSED, IN WHOLE OR IN PART, BY THE SOLE, JOINT, JOINT AND SEVERAL, CONCURRENT, CONTRIBUTORY, ACTIVE OR PASSIVE NEGLIGENCE OR THE STRICT LIABILITY OR FAULT OF THE PARTY BEING RELEASED OR INDEMNIFIED.

5.7 <u>Remedies Cumulative</u>. The remedies provided in this Article V shall be cumulative and shall not preclude assertion by any Indemnitee of any other rights or the seeking of any and all other remedies against any Indemnifying Party; <u>provided</u>, <u>however</u>, if a party has recovered any Losses from the other party pursuant to any provision of this Agreement or any Ancillary Agreement or otherwise, it shall not be entitled to recover the same Losses pursuant to any other provision of this Agreement or any Ancillary Agreement or otherwise.

5.8 <u>Survival of Indemnities</u>. The rights and obligations of each of Valero and Corner Store and their respective Indemnitees under this Article V shall survive (a) the sale or other transfer by any party of any Assets or businesses or the assignment by it of any Liabilities, and (b) any merger, consolidation business combination, sale of all or substantially all of the Assets, restructuring, recapitalization, reorganization or similar transaction involving either party or any of its respective Subsidiaries.

5.9 <u>Guarantees, Letters of Credit and Other Obligations</u>. In furtherance of, and not in limitation of, the obligations set forth in Section 2.6:

(a) On or prior to the Distribution Date or as soon as practicable thereafter, Corner Store shall (with the reasonable cooperation of the applicable member(s) of the Valero Group) use its commercially reasonable efforts to have any member(s) of the Valero Group removed as guarantor of or obligor for any Corner Store Liability to the extent that they relate to Corner Store Liabilities, including in respect of those guarantees, letters of credit and other obligations set forth on <u>Schedule 5.9(a)</u>.

(b) On or prior to the Distribution Date, to the extent required to obtain a release from a guarantee, letter of credit or other obligation of any member of the Valero Group, Corner Store shall execute a substitute document in the form of any such existing guarantee or letter of credit, as applicable, or such other form as is agreed to by the relevant parties to such guarantee agreement, letter of credit or other obligation, except to the extent that such existing guarantee contains representations, covenants or other terms or provisions either (i) with which Corner Store would be reasonably unable to comply or (ii) which would be reasonably expected to be breached.

(c) If the parties are unable to obtain, or to cause to be obtained, any such required removal as set forth in clauses (a) and (b) of this Section 5.9, (i) Corner Store shall, and

shall cause the other members of the Corner Store Group to, indemnify, defend and hold harmless each of the Valero Indemnitees for any Liability arising from or relating to such guarantee, letter of credit or other obligation, as applicable, and shall, as agent or subcontractor for the applicable Valero Group guarantor or obligor, pay, perform and discharge fully all of the obligations or other Liabilities of such guarantor or obligor thereunder, and (ii) Corner Store shall not, and shall cause the other members of the Corner Store Group not to, agree to renew or extend the term of, increase any obligations under, or transfer to a third Person, any loan, guarantee, letter of credit, lease, contract or other obligation for which a member of the Valero Group is or may be liable unless all obligations of the members of the Valero Group with respect thereto are thereupon terminated by documentation satisfactory in form and substance to Valero in its sole and absolute discretion.

5.10 Right of Contribution.

(a) *Contribution.* If any right of indemnification contained in this Article V is held unenforceable or is unavailable for any reason, or is insufficient to hold harmless an Indemnitee in respect of any Liability for which such Indemnitee is entitled to indemnification hereunder, then the Indemnifying Party shall contribute to the amounts paid or payable by the Indemnitees as a result of such Liability (or actions in respect thereof) in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party and its Subsidiaries, on the one hand, and the Indemnitees entitled to contribution, on the other hand, as well as any other relevant equitable considerations.

(b) Allocation of Relative Fault. Solely for purposes of determining relative fault pursuant to this Section 5.10: (i) any fault associated with the business conducted with the Corner Store Assets or Corner Store Liabilities (except for the gross negligence or intentional misconduct of Valero or a Valero Subsidiary) or with the ownership, operation or activities of the Corner Store Business prior to the Distribution Date, shall be deemed to be the fault of Corner Store and the Corner Store Subsidiaries, and no such fault shall be deemed to be the fault of Valero or a Valero Subsidiary; and (ii) any fault associated with the business conducted with the Excluded Assets or Excluded Liabilities (except for the gross negligence or intentional misconduct of Corner Store or a Corner Store Subsidiary) or with the ownership, operation or activities of the Valero Business prior to the Distribution Date, shall be deemed to be the fault of Valero and the Valero Business prior to the Distribution Date, shall be deemed to be the fault of Valero and the Valero Business prior to the Distribution Date, shall be deemed to be the fault of Valero and the Valero Business prior to the Distribution Date, shall be deemed to be the fault of Valero and the Valero Subsidiaries, and no such fault shall be deemed to be the fault of Valero and the Valero Subsidiaries, and no such fault shall be deemed to be the fault of Valero and the Valero Subsidiaries, and no such fault shall be deemed to be the fault of Corner Store or a Corner Store Subsidiaries, and no such fault shall be deemed to be the fault of Valero and the Valero Subsidiaries, and no such fault shall be deemed to be the fault of Corner Store or a Corner Store Subsidiary.

(c) *Contribution Procedures.* The provisions of Sections 5.5 and 5.6 shall govern any contribution claims.

5.11 <u>No Impact on Third Parties</u>. For the avoidance of doubt, except as expressly set forth in this Agreement, the indemnifications provided for in this Article V are made only for purposes of allocating responsibility for Liabilities between the Valero Group, on the one hand, and the Corner Store Group, on the other hand, and are not intended to, and shall not, affect any obligations to, or give rise to any rights of, any third parties.

5.12 <u>No Cross-Claims or Third-Party Claims</u>. Each of Corner Store and Valero agrees that it shall not, and shall not permit the members of its respective Group to, in connection with

any Third-Party Claim, assert as a counterclaim or third-party claim against any member of the Valero Group or Corner Store Group, respectively, any claim (whether sounding in contract, tort or otherwise) that arises out of or relates to this Agreement, any breach or alleged breach hereof, the transactions contemplated hereby (including all actions taken in furtherance of the transactions contemplated hereby on or prior to the date hereof), or the construction, interpretation, enforceability or validity hereof, which in each such case shall be asserted only as contemplated by Article IV.

5.13 <u>Severability</u>. If any indemnification provided for in this Article V is determined by a Texas federal or state court to be invalid, void or unenforceable, the liability shall be apportioned between the Indemnitee and the Indemnifying Party as determined in a separate proceeding in accordance with Article IV.

5.14 Ancillary Agreements. Notwithstanding anything in this Agreement to the contrary, to the extent (a) any Ancillary Agreement contains any indemnification obligation or contribution obligation relating to any Corner Store Liability, Excluded Liability, Corner Store Asset or Excluded Asset contributed, assumed, retained, transferred, delivered, conveyed or governed pursuant to such Ancillary Agreement, the indemnification obligations and contribution obligations contained herein shall not apply to such Corner Store Liability, Excluded Liability, Corner Store Asset or Excluded Asset and instead the indemnification obligations and/or contribution obligations set forth in such Ancillary Agreement shall govern with regard to such Corner Store Liability, Excluded Liability, Corner Store Asset or Excluded Asset, and (b) the Registration Rights Agreement contains any indemnification obligation or contribution obligation relating to any Liabilities of the Valero Indemnitees or Corner Store Indemnitees relating to, arising out of or resulting from information contained in any registration statement (other than the Form 10 (including any amendments or supplements thereto) and the Information Statement (as amended or supplemented if Corner Store shall have furnished any amendments or supplements thereto), to which this Section 5.14 shall not apply), the indemnification obligations and contribution obligations contained herein shall not apply to such Liabilities and instead the indemnification obligations and/or contribution obligations set forth in the Registration Rights Agreement shall govern with regard to such Liabilities.

ARTICLE VI INSURANCE MATTERS

6.1 Insurance Matters.

(a) Valero and Corner Store agree to cooperate in good faith to arrange insurance coverage for Corner Store to be effective no later than the Distribution Date. In no event shall Valero, any other member of the Valero Group or any Valero Indemnitee have liability or obligation whatsoever to any member of the Corner Store Group in the event that any insurance policy or other contract or policy of insurance shall be terminated or otherwise cease to be in effect for any reason, shall be unavailable or inadequate to cover any Liability of any member of the Corner Store Group for any reason whatsoever or shall not be renewed or extended beyond the current expiration date. (b) From and after the Distribution Date, neither Corner Store nor any member of the Corner Store Group shall have any rights to or under any of Valero's or its Affiliates' insurance policies. At the Distribution Date, Corner Store shall have in effect all insurance programs required to comply with Corner Store's contractual obligations and such other insurance policies as reasonably necessary, and, following the Distribution Date, Corner Store shall maintain such insurance programs and policies with insurers which comply with the minimum financial credit rating standards set by the major global insurance brokers.

(c) From and after the Distribution Date with respect to any losses, damages and liabilities incurred by any member of the Corner Store Group prior to or in respect of the period prior to the Distribution Date, Valero shall provide Corner Store with access to, and Corner Store may, upon ten (10) days' prior written notice to Valero, make claims under, Valero's third-party insurance policies and captive insurance policies in place at the time of the Distribution and Valero's historical policies of insurance, but solely to the extent that such policies provided coverage for the Corner Store Group prior to the Distribution; <u>provided</u>, that such access to, and the right to make claims under such insurance policies, shall be subject to the terms and conditions of such insurance policies, including any limits on coverage or scope, any deductibles, self-insured retentions and other fees and expenses, and shall be subject to the following additional conditions:

(i) Corner Store shall provide Valero with a written report sixty (60) days prior to any such third-party insurance policy's renewal date, as advised by Valero, identifying any claims made by Corner Store for which notice has previously been provided to insurers of Valero;

(ii) Corner Store and members of the Corner Store Group shall indemnify, hold harmless and reimburse Valero and members of the Valero Group for any deductibles, self-insured retention, fees and expenses incurred by Valero or members of the Valero Group to the extent resulting from any such access to, or any claims made by Corner Store or members of the Corner Store Group under, any insurance provided pursuant to this Section 6.1(c), including any indemnity payments, settlements, judgments, legal fees and allocated claims expenses and claim-handling fees, whether such claims are made by Corner Store, its employees or third Persons; and

(iii) Corner Store shall exclusively bear (and neither Valero nor any member of the Valero Group shall have any obligation to repay or reimburse Corner Store or members of the Corner Store Group for) and shall be liable for all uninsured, uncovered, unavailable or uncollectible amounts of all such claims made by Corner Store or any member of the Corner Store Group under the policies as provided for in this Section 6.1(c).

(iv) Notwithstanding the foregoing, nothing in this Section 6.1(c) shall be construed to limit any of Valero's indemnification obligations, or any Corner Store Indemnitee's rights to receive indemnification, under Article V.

In the event that an insurance policy aggregate is exhausted, or believed likely to be exhausted, due to noticed claims, the Corner Store Group, on the one hand, and the Valero Group, on the

other hand, shall be responsible for their *pro rata* portion of the reinstatement premium, based upon the losses of such Group submitted to Valero's insurance carrier(s) (including any submissions prior to the Distribution Date). To the extent that the Valero Group or the Corner Store Group is allocated more than its *pro rata* portion of such premium due to the timing of losses submitted to Valero's insurance carrier(s), the other party shall promptly pay the first party an amount so that each Group has been properly allocated its *pro rata* portion of the reinstatement premium. If Valero determines not to reinstate the policy aggregate, then each Group shall bear all of its own future costs.

In the event that any member of the Valero Group incurs any losses, damages or liability prior to or in respect of the period prior to the Distribution Date for which such member of the Valero Group is entitled to coverage under Corner Store's third-party insurance policies, the same process pursuant to this Section 6.1(c) shall apply, substituting "Valero" for "Corner Store" and "Corner Store" for "Valero."

(d) All payments and reimbursements by Corner Store pursuant to this Section 6.1 shall be made within fifteen (15) days after Corner Store's receipt of an invoice therefor from Valero. If Valero incurs costs to enforce Corner Store's obligations herein, Corner Store agrees to indemnify Valero for such enforcement costs, including attorneys' fees.

All payments and reimbursements by Valero pursuant to this Section 6.1 (e) shall be made within fifteen (15) days after Valero's receipt of an invoice therefor from Corner Store. If Corner Store incurs costs to enforce Valero's obligations herein, Valero agrees to indemnify Corner Store for such enforcement costs, including attorneys' fees. Valero shall retain the exclusive right to control its insurance policies and programs, including the right to exhaust. settle, release, commute, buy back or otherwise resolve disputes with respect to any of its insurance policies and programs and to amend, modify or waive any rights under any such insurance policies and programs, notwithstanding whether any such policies or programs apply to any Corner Store Liabilities and/or claims Corner Store has made or could make in the future. and no member of the Corner Store Group shall, without the prior written consent of Valero, erode, exhaust, settle, release, commute, buy-back or otherwise resolve disputes with Valero's insurers with respect to any of Valero's insurance policies and programs, or amend, modify or waive any rights under any such insurance policies and programs. Corner Store shall cooperate with Valero and share such information at Corner Store's cost as is reasonably necessary in order to permit Valero to manage and conduct its insurance matters as it deems appropriate. Neither Valero nor any of its Affiliates shall have any obligation to secure extended reporting for any claims under any of Valero's or its Affiliates' liability policies for any acts or omissions by any member of the Corner Store Group incurred prior to the Distribution Date. To the extent reasonably practicable, Valero will notify Corner Store at least 10 days prior to terminating or initiating a buyback of any insurance policy or program with respect to which Corner Store has asserted a claim or given written notice to Valero that it proposes to submit a claim.

(f) This Agreement shall not be considered as an attempted assignment of any policy of insurance or as a contract of insurance and shall not be construed to waive any right or remedy of any member of the Valero Group in respect of any insurance policy or any other contract or policy of insurance.

(g) Corner Store does hereby, for itself and each other member of the Corner Store Group, agree that no member of the Valero Group shall have any Liability whatsoever as a result of the insurance policies and practices of Valero and its Affiliates as in effect at any time, including as a result of the level or scope of any such insurance, the creditworthiness of any insurance carrier, the terms and conditions of any policy, or the adequacy or timeliness of any notice to any insurance carrier with respect to any claim or potential claim or otherwise.

ARTICLE VII

EXCHANGE OF INFORMATION; CONFIDENTIALITY

7.1 Agreement for Exchange of Information. Except as otherwise provided in any Ancillary Agreement, each of Valero and Corner Store, on behalf of itself and the members of its respective Group, shall use commercially reasonable efforts to provide or make available, or cause to be provided or made available, to the other party, at any time before or after the Distribution Date, as soon as reasonably practicable after written request therefor, any Information (or a copy thereof) in the possession or under the control of either party or any of its Subsidiaries to the extent that: (i) such Information relates to the Corner Store Business or any Corner Store Asset or Corner Store Liability, if Corner Store is the requesting party, or to the Valero Business or any Excluded Asset or Excluded Liability, if Valero is the requesting party; (ii) such Information is required by the requesting party to comply with its obligations under this Agreement or any Ancillary Agreement; or (iii) such Information is required by the requesting party to comply with any obligation imposed by any Governmental Authority; provided, however, that, in the event that the party to whom the request has been made determines that any such provision of Information could be commercially detrimental, violate any Law or agreement or waive any attorney-client privilege, then the parties shall use commercially reasonable efforts to permit compliance with such obligations to the extent and in a manner that avoids any harm or consequence. The party providing Information pursuant to this Section 7.1 shall only be obligated to provide such Information in the form, condition and format in which it then exists and in no event shall such party be required to perform any improvement, modification, conversion, updating or reformatting of any such Information, and nothing in this Section 7.1 shall expand the obligations of the parties under Section 7.4.

7.2 <u>Ownership of Information</u>. Any Information owned by one Group that is provided to a requesting party pursuant to Section 7.1 or 7.7 shall remain the property of the providing party. Unless specifically set forth herein, nothing contained in this Agreement shall be construed as granting or conferring rights of license or otherwise in any such Information.

7.3 <u>Compensation for Providing Information</u>. The party requesting Information agrees to reimburse the other party for the reasonable costs, if any, of creating, gathering and copying such Information or otherwise complying with the request with respect to such Information. Except as may be otherwise specifically provided elsewhere in this Agreement or in any other agreement between the parties, such costs shall be computed in accordance with the providing party's standard methodology and procedures.

7.4 <u>Record Retention</u>.

(a) The parties agree and acknowledge that it is not practicable to separate all Tangible Information belonging to the parties, and that following the Distribution Date, each party will have some of the Tangible Information of the other party stored at its facilities or at Third Party records storage locations arranged for by such party (each, a "<u>Records Facility</u>").

(b) Each party shall use commercially reasonable efforts: (i) to maintain the Stored Records as to which it is the Custodial Party in accordance with its regular records retention policies and procedures and the terms of this Section 7.4; and (ii) to comply with the requirements of any "litigation hold" that relates to Stored Records as to which it is the Custodial Party that relates to (x) any Action that is pending as of the Distribution Date or (y) any Action that arises or becomes threatened or reasonably anticipated after the Distribution Date as to which the Custodial Party has received a written notice of the applicable "litigation hold" from the Non-Custodial Party.

7.5 <u>Limitations of Liability</u>. No party shall have any liability to any other party in the event that any Information exchanged or provided pursuant to this Agreement is found to be inaccurate in the absence of willful misconduct by the party providing such Information. No party shall have any liability to any other party if any Information is destroyed after commercially reasonable efforts by such party to comply with the provisions of Section 7.4.

7.6 Other Agreements Providing for Exchange of Information.

(a) The rights and obligations granted under this Article VII are subject to any specific limitations, qualifications or additional provisions on the sharing, exchange, retention or confidential treatment of Information set forth herein or any Ancillary Agreement.

(b) Either party that receives, pursuant to a request for Information in accordance with this Article VII, Tangible Information that is not relevant to its request shall (i) return it to the providing party or, at the providing party's request, destroy such Tangible Information and (ii) deliver to the providing party a certificate certifying that such Tangible Information was returned or destroyed, as the case may be, which certificate shall be signed by an authorized Representative of the requesting party.

(c) When any Tangible Information provided by one party to the other party (other than Tangible Information provided pursuant to Section 7.4) is no longer needed for the purposes contemplated by this Agreement or any Ancillary Agreement or is no longer required to be retained by applicable Law, the receiving party shall promptly, after request of the other party, either return to the other party all Tangible Information in the form in which it was originally provided (including all copies thereof and all notes, extracts or summaries based thereon) or, if the providing party has requested that the other party destroy such Tangible Information, certify to the other party that it has destroyed such Tangible Information (and such copies thereof and such notes, extracts or summaries based thereon); <u>provided</u>, that this obligation to return or destroy such Tangible Information shall not apply to any Tangible Information solely related to the receiving party's business, Assets, Liabilities, operations or activities.

7.7 <u>Production of Witnesses; Records; Cooperation</u>. Without limiting any other provision of this Agreement, the parties agree to consult and cooperate to the extent reasonably necessary with respect to any Actions. In furtherance of such undertaking, the parties shall enter into a mutually acceptable joint defense agreement so as to maintain to the extent practicable any applicable attorney-client privilege or work product immunity of any member of any Group.

7.8 <u>Privileged Matters</u>.

(a) The parties recognize that legal and other professional services that have been and shall be provided prior to the Distribution Date have been and shall be rendered for the collective benefit of the parties and their respective Subsidiaries, and that each party and its respective Subsidiaries should be deemed to be the client with respect to such services for the purposes of asserting all privileges and immunities that may be asserted under applicable Law in connection therewith.

(b) The parties agree as follows: (i) Valero shall be entitled, in perpetuity, to control the assertion or waiver of all privileges and immunities in connection with any Privileged Information that relates solely to the Valero Business, whether or not the Privileged Information is in the possession or under the control of a member of the Valero Group or the Corner Store Group. Valero shall also be entitled, in perpetuity, to control the assertion or waiver of all privileges and immunities in connection with any Privileged Information that relates solely to any Excluded Liabilities resulting from any Actions that are now pending or may be asserted in the future, whether or not the Privileged Information is in the possession or under the control of a member of the Valero Group or the Corner Store Group; and (ii) Corner Store shall be entitled, in perpetuity, to control the assertion or waiver of all privileges and immunities in connection with any Privileged Information that relates solely to the Corner Store Business, whether or not the Privileged Information is in the possession or under the control of a member of the Valero Group or the Corner Store Group. Corner Store shall also be entitled, in perpetuity, to control the assertion or waiver of all privileges and immunities in connection with any Privileged Information that relates solely to any Corner Store Liabilities resulting from any Actions that are now pending or may be asserted in the future, whether or not the Privileged Information is in the possession or under the control of a member of the Valero Group or the Corner Store Group.

(c) Subject to Sections 7.8(d) and 7.8(e), the parties agree that they shall have a shared privilege or immunity with respect to all privileges not allocated pursuant to Section 7.8(b) and all privileges and immunities relating to any Actions or other matters that involve both parties (or one or more of their respective Subsidiaries) and in respect of which both parties have Liabilities under this Agreement, and that no such shared privilege or immunity may be waived by either party without the consent of the other party.

(d) If any dispute arises between Valero and Corner Store, or any of their respective Subsidiaries, regarding whether a privilege or immunity should be waived to protect or advance the interests of either party and/or their respective Subsidiaries, each party agrees that it shall: (i) negotiate with the other party in good faith; (ii) endeavor to minimize any prejudice to the rights of the other party; and (iii) not unreasonably withhold consent to any request for waiver by the other party. Further, each party specifically agrees that it shall not withhold its

consent to the waiver of a privilege or immunity for any purpose except to protect its own legitimate interests.

(e) Upon receipt by Corner Store or by any of the Corner Store Subsidiaries of any subpoena, discovery or other request that may reasonably be expected to result in the production or disclosure of Information subject to a shared privilege or immunity or as to which Valero or any of the Valero Subsidiaries has the sole right hereunder to assert a privilege or immunity, or if Corner Store obtains knowledge that any of its, or a Corner Store Subsidiary's, current or former directors, officers, agents or employees have received any subpoena, discovery or other requests that may reasonably be expected to result in the production or disclosure of such Privileged Information, Corner Store shall promptly provide written notice to Valero of the existence of the request (which notice shall be delivered to Valero no later than five (5) business days following the receipt of any such subpoena, discovery or other request) and shall provide Valero a reasonable opportunity to review the Information and to assert any rights it or they may have, including under this Section 7.8 or otherwise, to prevent the production or disclosure of such Privileged Information.

(f) Upon receipt by Valero or by any of the Valero Subsidiaries of any subpoena, discovery or other request that may reasonably be expected to result in the production or disclosure of Information subject to a shared privilege or immunity or as to which Corner Store or any of the Corner Store Subsidiaries has the sole right hereunder to assert a privilege or immunity, or if Valero obtains knowledge that any of its, or a Valero Subsidiary's, current or former directors, officers, agents or employees have received any subpoena, discovery or other requests that may reasonably be expected to result in the production or disclosure of such Privileged Information, Valero shall promptly provide written notice to Corner Store of the existence of the request (which notice shall be delivered to Corner Store no later than five (5) business days following the receipt of any such subpoena, discovery or other request) and shall provide Corner Store a reasonable opportunity to review the Information and to assert any rights it or they may have, including under this Section 7.8 or otherwise, to prevent the production or disclosure of such Privileged Information.

(g) Any furnishing of, or access to, Information pursuant to this Agreement is made in reliance on the agreement of the parties set forth in this Section 7.8 and in Section 7.9 to maintain the confidentiality of Privileged Information and to assert and maintain all applicable privileges and immunities. The parties further agree that: (i) the exchange by one party to the other party of any Privileged Information that should not have been transferred pursuant to the terms of this Article VII shall not be deemed to constitute a waiver of any privilege or immunity that has been or may be asserted under this Agreement or otherwise with respect to such Privileged Information; and (ii) the party receiving such Privileged Information shall promptly return such Privileged Information to the party who has the right to assert the privilege or immunity.

(h) In furtherance of, and without limitation to, the parties' agreement under this Section 7.8, Valero and Corner Store shall, and shall cause their applicable Subsidiaries to, use reasonable efforts to maintain their respective separate and joint privileges and immunities, including by executing joint defense and/or common interest agreements where necessary or useful for this purpose.

7.9 <u>Confidentiality</u>.

Confidentiality. From and after the Distribution Date, subject to Section (a) 7.10 and except as contemplated by or otherwise provided in this Agreement or any Ancillary Agreement, Valero, on behalf of itself and each of the Valero Subsidiaries, and Corner Store, on behalf of itself and each of the Corner Store Subsidiaries, agrees to hold, and to cause its respective Representatives to hold, in strict confidence, with at least the same degree of care that applies to Valero's confidential and proprietary information pursuant to policies in effect as of the Distribution Date, all confidential and proprietary Information concerning the other party (or its business) and the other party's Subsidiaries (or their respective businesses) that is either in its possession (including confidential and proprietary Information in its possession prior to the Distribution Date) or furnished by the other party or the other party's Subsidiaries or their respective Representatives at any time pursuant to this Agreement or any Ancillary Agreement, and shall not use any such confidential and proprietary Information other than for such purposes as may be expressly permitted hereunder or thereunder, except, in each case, to the extent that such confidential and proprietary Information has been: (i) in the public domain or generally available to the public, other than as a result of a disclosure by such party or any of its Subsidiaries or any of their respective Representatives in violation of this Agreement; (ii) later lawfully acquired from other sources by such party or any of its Subsidiaries, which sources are not themselves bound by a confidentiality obligation or other contractual, legal or fiduciary obligation of confidentiality with respect to such confidential and proprietary Information; or (iii) independently developed or generated without reference to or use of the respective proprietary or confidential Information of the other party or any of its Subsidiaries. If any confidential and proprietary Information of one party or any of its Subsidiaries is disclosed to another party or any of its Subsidiaries in connection with providing services to such first party or any of its Subsidiaries under this Agreement or any Ancillary Agreement, then such disclosed confidential and proprietary Information shall be used only as required to perform such services.

No Release; Return or Destruction. Each party agrees not to release or (b)disclose, or permit to be released or disclosed, any confidential or proprietary Information of the other party addressed in Section 7.9(a) to any other Person, except its Representatives who need to know such Information in their capacities as such, and except in compliance with Section 7.10. Without limiting the foregoing, when any Information furnished by the other party after the Distribution Date pursuant to this Agreement or any Ancillary Agreement is no longer needed for the purposes contemplated by this Agreement or any Ancillary Agreement, each party shall, at its option, promptly after receiving a written notice from the disclosing party, either return to the disclosing party all such Information in a tangible form (including all copies thereof and all notes, extracts or summaries based thereon) or certify to the disclosing party that it has destroyed such Information (and such copies thereof and such notes, extracts or summaries based thereon); provided, however, that a party shall not be required to destroy or return any such Information to the extent that (i) the party is required to retain the Information in order to comply with any applicable Law, (ii) the Information has been backed up electronically pursuant to the party's standard document retention policies and will be managed and ultimately destroyed consistent with such policies or (iii) it is kept in the party's legal files for purposes of resolving any dispute that may arise under this Agreement or any Ancillary Agreement.

(c) Third-Party Information; Privacy or Data Protection Laws. Each party acknowledges that it and its respective Subsidiaries may presently have and, following the Distribution Date, may gain access to or possession of confidential or proprietary Information of, or personal Information relating to, Third Parties: (i) that was received under confidentiality or non-disclosure agreements entered into between such Third Parties, on the one hand, and the other party or the other party's Subsidiaries, on the other hand, prior to the Distribution Date; or (ii) that, as between the two parties, was originally collected by the other party or the other party's Subsidiaries and that may be subject to and protected by privacy, data protection or other applicable Laws. As may be provided in more detail in an applicable Ancillary Agreement, each party agrees that it shall hold, protect and use, and shall cause its Subsidiaries and its and their respective Representatives to hold, protect and use, in strict confidence the confidential and proprietary Information of, or personal Information relating to, Third Parties in accordance with privacy, data protection or other applicable Laws and the terms of any agreements that were either entered into before the Distribution Date or affirmative commitments or representations that were made before the Distribution Date by, between or among the other party or the other party's Subsidiaries, on the one hand, and such Third Parties, on the other hand.

Protective Arrangements. In the event that either party or any of its Subsidiaries 7.10 is requested or required (by oral question, interrogatories, requests for information or documents, subpoena, civil investigative demand or similar process) by any Governmental Authority or pursuant to applicable Law or the rules of any stock exchange on which the shares of the party or any member of its Group are traded to disclose or provide any confidential or proprietary Information of the other party (other than with respect to any such Information furnished pursuant to the provisions of Sections 7.1 through 7.9, as applicable), that is subject to the confidentiality provisions hereof, such party shall provide the other party with written notice of such request or demand as promptly as practicable under the circumstances so that such other party shall have an opportunity to seek an appropriate protective order, at such other party's own cost and expense. In the event that such other party fails to receive such appropriate protective order in a timely manner and the party receiving the request or demand reasonably determines that its failure to disclose or provide such Information shall actually prejudice the party receiving the request or demand, then the party that received such request or demand may thereafter disclose or provide Information to the extent required by such Law (as so advised by counsel) or by lawful process or such Governmental Authority.

ARTICLE VIII FURTHER ASSURANCES AND ADDITIONAL COVENANTS

8.1 <u>Further Assurances</u>.

(a) In addition to the actions specifically provided for elsewhere in this Agreement, each of the parties hereto shall use its commercially reasonable efforts, prior to, on and after the Distribution Date, to take, or cause to be taken, all actions, and to do, or cause to be done, all things reasonably necessary, proper or advisable on its part under applicable Laws, regulations and agreements, to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements.

(b)Without limiting the foregoing, prior to, on and after the Distribution Date, each party hereto shall cooperate with each other party hereto, and without any further consideration, but at the expense of the requesting party, to execute and deliver, or use its commercially reasonable efforts to cause to be executed and delivered, all instruments, including instruments of conveyance, assignment and transfer, and to make all filings with, and to obtain or make any Approvals or Notifications of, any Governmental Authority or any other Person under any permit, license, agreement, indenture or other instrument (including any Third-Party consents or Governmental Approvals), and to take all such other actions as such party may reasonably be requested to take by any other party hereto from time to time, consistent with the terms of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements and the transfers of the Corner Store Assets and the assignment and assumption of the Corner Store Liabilities and the other transactions contemplated hereby and thereby. Without limiting the foregoing, each party shall, at the reasonable request, cost and expense of any other party, take such other actions as may be reasonably necessary to vest in such other party all of the transferring party's right, title and interest to the Assets allocated to such party by this Agreement or any Ancillary Agreement, in each case, if and to the extent it is practicable to do so.

(c) On or prior to the Distribution Date, Valero and Corner Store in their respective capacities as direct and indirect stockholders of their respective Subsidiaries, shall each ratify any actions that are reasonably necessary or desirable to be taken by any Subsidiary of Valero or Subsidiary of Corner Store, as the case may be, to effectuate the transactions contemplated by this Agreement and the Ancillary Agreements.

8.2 <u>Performance</u>. Valero shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the Valero Group. Corner Store shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the Corner Store Group. Each party (including its permitted successors and assigns) further agrees that it shall (a) give timely notice of the terms, conditions and continuing obligations contained in this Section 8.2 to all of the other members of its Group, and (b) cause all of the other members of its Group not to take, or omit to take, any action which action or omission would violate or cause such party to violate this Agreement or any Ancillary Agreement.

8.3 Order of Precedence. Notwithstanding anything to the contrary in this Agreement or any Ancillary Agreement, in the case of any conflict between the provisions of this Agreement and the provisions of any Ancillary Agreement, the provisions of this Agreement shall prevail; <u>provided</u>, <u>however</u>, that in relation to any matters concerning Taxes, the Tax Matters Agreement shall prevail over this Agreement and any other Ancillary Agreement, and in relation to any matters governed by the Employee Matters Agreement, the Employee Matters Agreement shall prevail over this Agreement and any other Ancillary Agreement.

ARTICLE IX TERMINATION

9.1 <u>Termination</u>. This Agreement and any Ancillary Agreement may be terminated and the terms and conditions of the Separation and the Distribution may be amended, modified or abandoned at any time prior to the Distribution Date by and in the sole and absolute discretion of the Valero Board without the approval of any Person, including Corner Store. In the event that this Agreement is terminated, this Agreement shall become null and void and no party, nor any party's directors, officers or employees, shall have any liability of any kind to any Person by reason of this Agreement. After the Distribution, this Agreement may not be terminated except by an agreement in writing signed by each of the parties to this Agreement.

ARTICLE X MISCELLANEOUS

10.1 Counterparts; Entire Agreement; Corporate Power.

(a) This Agreement and each Ancillary Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each party and delivered to each other party.

(b) This Agreement and the Ancillary Agreements, and the exhibits, annexes and schedules hereto and thereto, contain the entire agreement between the parties with respect to the subject matter hereof, supersede all previous agreements, negotiations, discussions, writings, understandings, commitments and conversations with respect to such subject matter and there are no agreements or understandings between the parties with respect to such subject matter other than those set forth or referred to herein or therein.

(c) Valero represents on behalf of itself and each other member of the Valero Group, and Corner Store represents on behalf of itself and each other member of the Corner Store Group, as follows:

(i) each such Person has the requisite corporate or other power and authority and has taken all corporate or other action necessary in order to execute, deliver and perform this Agreement and each Ancillary Agreement to which it is a party and to consummate the transactions contemplated hereby and thereby; and

(ii) this Agreement and each Ancillary Agreement to which it is a party have been duly executed and delivered by it and constitutes a valid and binding agreement of it enforceable in accordance with the terms thereof.

(d) Each party hereto acknowledges that it and each other party hereto may execute this Agreement and the Ancillary Agreements by facsimile, stamp or mechanical signature. Each party hereto expressly adopts and confirms each such facsimile, stamp or mechanical signature made in its respective name as if it were a manual signature, agrees that it shall not assert that any such signature is not adequate to bind such party to the same extent as if it were signed manually and agrees that at the reasonable request of any other party hereto at any time it shall as promptly as reasonably practicable cause this Agreement and each Ancillary Agreement to be manually executed (any such execution to be as of the date of the initial date thereof).

10.2 <u>Governing Law</u>. This Agreement and, unless expressly provided therein, each Ancillary Agreement (and any claims or disputes arising out of or related hereto or thereto or to the transactions contemplated hereby and thereby or to the inducement of any party to enter herein and therein, whether for breach of contract, tortious conduct or otherwise and whether predicated on common law, statute or otherwise) shall be governed by and construed and interpreted in accordance with the Laws of the State of Texas, irrespective of the choice of laws principles of the State of Texas, including all matters of validity, construction, effect, enforceability, performance and remedies.

10.3 <u>Assignability</u>. Except as set forth in any Ancillary Agreement, this Agreement and each Ancillary Agreement shall be binding upon and inure to the benefit of the parties hereto and thereto, respectively, and their respective successors and permitted assigns; <u>provided</u>, <u>however</u>, that no party hereto or thereto may assign its respective rights or delegate its respective obligations under this Agreement or any Ancillary Agreement without the express prior written consent of the other parties hereto or thereto and <u>provided</u>, <u>further</u>, that Valero may assign its rights and delegate its obligations under this Agreement or any Ancillary Agreement to any of its Affiliates or in connection with a change of control of Valero or a sale or disposition of any of the Assets or Liabilities or lines of business of Valero, <u>provided</u>, that no such assignment or delegation shall release Valero from any Liability or obligation under this Agreement or any Ancillary Agreement.

10.4 <u>Third-Party Beneficiaries</u>. Except for the release and indemnification rights under this Agreement or any Ancillary Agreement of any Valero Indemnitee or Corner Store Indemnitee in their respective capacities as such, (a) the provisions of this Agreement and each Ancillary Agreement are solely for the benefit of the parties and are not intended to confer upon any Person except the parties any rights or remedies hereunder or thereunder; and (b) there are no third-party beneficiaries of this Agreement or any Ancillary Agreement and neither this Agreement nor any Ancillary Agreement shall provide any third Person with any remedy, claim, liability, reimbursement, claim of action or other right in excess of those existing without reference to this Agreement or any Ancillary Agreement.

10.5 <u>Notices</u>. All notices, requests, claims, demands or other communications under this Agreement and, to the extent applicable and unless otherwise provided therein, under each of the Ancillary Agreements, shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by facsimile or electronic transmission with receipt confirmed (followed by delivery of an original via overnight courier service), or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified in a notice given in accordance with this Section 10.5): If to Valero, to:

Valero Energy Corporation One Valero Way San Antonio, Texas 78249 Attention: General Counsel

If to Corner Store, to:

CST Brands, Inc. One Valero Way San Antonio, Texas 78249 Attention: General Counsel

Any party may, by notice to the other party, change the address and contact person to which any such notices are to be given.

10.6 <u>Severability</u>. If any provision of this Agreement or any Ancillary Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof or thereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby. Upon such determination, the parties shall negotiate in good faith in an effort to agree upon such a suitable and equitable provision to effect the original intent of the parties.

10.7 Force Majeure. Neither party shall be deemed in default of this Agreement, the Tax Matters Agreement or the Employee Matters Agreement for failure to fulfill any obligation, other than a delay or failure to make a payment, so long as and to the extent to which any delay or failure in the fulfillment of such obligations is prevented, frustrated, hindered or delayed as a consequence of circumstances of Force Majeure. In the event of any such excused delay, the time for performance shall be extended for a period equal to the time lost by reason of the delay. A party claiming the benefit of this provision shall, as soon as reasonably practicable after the occurrence of any such event, (a) provide written notice to the other party of the nature and extent of any such Force Majeure condition; and (b) use commercially reasonable efforts to remove any such causes and resume performance under this Agreement, the Tax Matters Agreement or the Employee Matters Agreement, as applicable, as soon as reasonably practicable.

10.8 <u>Publicity</u>. From and after the Distribution Date, Corner Store and Valero shall consult with each other before issuing, and give each other the opportunity to review and comment upon, any press release or other public statements with respect to the transactions contemplated by this Agreement and the Ancillary Agreements, and shall not issue any such press release or make any public statement prior to such consultation, except as may be required by applicable Law, court process or by obligations pursuant to any listing agreement with any national securities exchange or national securities quotation system. 10.9 <u>Expenses</u>. Except as expressly set forth in this Agreement (including Sections 3.1(g) and 8.1(b)) or in any Ancillary Agreement, all fees, costs and expenses incurred in connection with the preparation, execution, delivery and implementation of this Agreement and any Ancillary Agreement, and with the consummation of the transactions contemplated hereby and thereby, shall be borne by the party incurring such fees, costs or expenses.

10.10 <u>Late Payments</u>. Except as expressly provided to the contrary in this Agreement or in any Ancillary Agreement, any amount not paid when due pursuant to this Agreement or any Ancillary Agreement (and any amounts billed or otherwise invoiced or demanded and properly payable that are not paid within sixty (60) days of such bill, invoice or other demand) shall accrue interest at a rate per annum equal to the Prime Rate plus two percent (2%).

10.11 <u>Headings</u>. The article, section and paragraph headings contained in this Agreement and in the Ancillary Agreements are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement or any Ancillary Agreement.

10.12 <u>Survival of Covenants</u>. Except as expressly set forth in any Ancillary Agreement, the covenants, representations and warranties contained in this Agreement and each Ancillary Agreement, and liability for the breach of any obligations contained herein or therein, shall survive the Separation and the Distribution and shall remain in full force and effect.

10.13 <u>Waivers of Default</u>. Waiver by any party of any default by the other party of any provision of this Agreement or any Ancillary Agreement shall not be deemed a waiver by the waiving party of any subsequent or other default, nor shall it prejudice the rights of such party. No failure or delay by any party in exercising any right, power or privilege under this Agreement or any Ancillary Agreement shall operate as a waiver thereof nor shall a single or partial exercise thereof prejudice any other or further exercise thereof or the exercise of any other right, power or privilege.

10.14 <u>Specific Performance</u>. Subject to the provisions of Article IV, in the event of any actual or threatened default in, or breach of, any of the terms, conditions and provisions of this Agreement or any Ancillary Agreement, the party or parties who are, or are to be, thereby aggrieved shall have the right to specific performance and injunctive or other equitable relief (on an interim or permanent basis) in respect of its or their rights under this Agreement or such Ancillary Agreement, in addition to any and all other rights and remedies at law or in equity, and all such rights and remedies shall be cumulative. The parties agree that the remedies at law for any breach or threatened breach, including monetary damages, are inadequate compensation for any loss and that any defense in any action for specific performance that a remedy at law would be adequate is waived. Any requirements for the securing or posting of any bond with such remedy are waived by each of the parties to this Agreement.

10.15 <u>Amendments</u>. No provisions of this Agreement or any Ancillary Agreement shall be deemed waived, amended, supplemented or modified by any party, unless such waiver, amendment, supplement or modification is in writing and signed by the authorized representative of the party against whom such waiver, amendment, supplement or modification is sought to be enforced.

10.16 Interpretation. In this Agreement and any Ancillary Agreement, (a) words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other genders as the context requires; (b) the terms "hereof," "herein," "herewith" and words of similar import, and the terms "Agreement" and "Ancillary Agreement" shall, unless otherwise stated, be construed to refer to this Agreement or the applicable Ancillary Agreement as a whole (including all of the Schedules, Exhibits and Appendices hereto and thereto) and not to any particular provision of this Agreement or such Ancillary Agreement; (c) Article, Section, Exhibit, Schedule and Appendix references are to the Articles, Sections, Exhibits, Schedules and Appendices to this Agreement (or the applicable Ancillary Agreement) unless otherwise specified; (d) the word "including" and words of similar import when used in this Agreement (or the applicable Ancillary Agreement) means "including, without limitation"; (e) the word "or" shall not be exclusive; (f) unless expressly stated to the contrary in this Agreement or in any Ancillary Agreement, all references to "the date hereof," "the date of this Agreement," "hereby" and "hereupon" and words of similar import shall all be references to the date first stated in the preamble to this Agreement, regardless of any amendment or restatement hereof; (g) unless otherwise provided, all references to "\$" or "dollars" are to United States dollars; and (h) references to the performance, discharge or fulfillment of any Liability in accordance with its terms shall have meaning only to the extent such Liability has terms, and if the Liability does not have terms, the reference shall mean performance, discharge or fulfillment of such Liability.

10.17 <u>Mutual Drafting</u>. This Agreement and the Ancillary Agreements shall be deemed to be the joint work product of the parties, and any rule of construction that a document shall be interpreted or construed against a drafter of such document shall not be applicable.

10.18 Limitations of Liability. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, NEITHER CORNER STORE OR ITS AFFILIATES, ON THE ONE HAND, NOR VALERO OR ITS AFFILIATES, ON THE OTHER HAND, SHALL BE LIABLE UNDER THIS AGREEMENT TO THE OTHER FOR ANY LOST PROFITS, LOST BUSINESS OPPORTUNITIES, OR ANY SPECIAL, INDIRECT, PUNITIVE, EXEMPLARY, REMOTE, SPECULATIVE OR SIMILAR DAMAGES IN EXCESS OF COMPENSATORY DAMAGES OF THE OTHER ARISING IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY (OTHER THAN ANY SUCH LIABILITY WITH RESPECT TO A THIRD-PARTY CLAIM). IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized representatives.

VALERO ENERGY CORPORATION

APPROVED LEGAL

William. By:

Name: William R. Klesse Title: Chief Executive Office

CST BRANDS, INC.

By:

Somes inte Name: Kimberly S. Bowers Title: Chief Executive officer and President

[Signature Page to Separation and Distribution Agreement]

Alameda County Environmental Cleanup	REVISION DATE: December 14, 2017			
Oversight Programs (LOP and SCP)	ISSUE DATE: July 25, 2012			
	PREVIOUS REVISIONS: September 17, 2013, May 15, 2014, December 12, 2016			
SECTION: ACDEH Procedures	SUBJECT: Responsible Party(ies) Legal Requirements / Obligations			

REPORT & DELIVERABLE REQUESTS

Alameda County Department of Environmental Health (ACDEH) Cleanup Oversight Programs, Local Oversight Program (LOP) and Site Cleanup Program (SCP) require submission of all reports in electronic form to the State Water Board's (SWB) GeoTracker website in accordance with California Code of Regulations, Chapter 30, Division3, Title 23 and Division 3, Title 27.

Leaking Underground Fuel Tank (LUFT) Cases

Reports and deliverable requests are pursuant to California Health and Safety Code Section 25296.10. 23 CCR Sections 2652 through 2654, and 2721 through 2728 outline the responsibilities of a responsible party (RP) in conjunction with an unauthorized release from a petroleum underground storage tank (UST) system.

Site Cleanup Program (SCP) Cases

For non-petroleum UST cases, reports and deliverables requests are pursuant to California Health and Safety Code Section 101480.

ELECTRONIC SUBMITTAL OF REPORTS

A complete report submittal includes the PDF report and all associated electronic data files, including but not limited to GEO_MAP, GEO_XY, GEO_Z, GEO_BORE, GEO_WELL, and laboratory analytical data in Electronic Deliverable Format[™] (EDF). Additional information on these requirements is available on the State Water Board's website (http://www.waterboards.ca.gov/water issues/programs/ust/electronic submittal/)

- Do not upload draft reports to GeoTracker
- Rotate each page in the PDF document in the direction that will make it easiest to read on a computer monitor.

GEOTRACKER UPLOAD CERTIFICATION

Each report submittal is to include a GeoTracker Upload Summary Table with GeoTracker valid values¹ as illustrated in the example below to facilitate ACDEH review and verify compliance with GeoTracker requirements.

GeoTracker Upload Table Example

Report Title	Sampl e Period	PDF Report	GEO_ MAPS	Sample ID	Matrix	GEO _Z	GEO _XY	GEO_ BORE	GEO_WEL L	EDF
2016 Subsurface Investigation Report	2016 S1	~	•	Effluent	SO					✓
2012 Site Assessment Work Plan	2012	~	~							
2010 GW Investigation	2008 Q4	✓	✓	SB-10	W	~				✓
Report				SB-10-6	SO					✓
				MW-1	WG	✓	✓	✓	✓	✓
				SW-1	W	~	~	✓	✓	✓

¹ GeoTracker Survey XYZ, Well Data, and Site Map Guidelines & Restrictions, CA State Water Resources Control Board, April 2005

Alameda County Environmental Cleanup Oversight Programs	REVISION DATE: NA		
	ISSUE DATE: December 14, 2017		
(LOP and SCP)	PREVIOUS REVISIONS: September 17, 2013, May 15, 2014, December 12, 2016		
SECTION: ACDEH Procedures	SUBJECT: Responsible Party(ies) Legal Requirements / Obligations		

ACKNOWLEDGEMENT STATEMENT

All work plans, technical reports, or technical documents submitted to ACDEH must be accompanied by a cover letter from the responsible party that states, at a minimum, the following: "I have read and acknowledge the content, recommendations and/or conclusions contained in the attached document or report submitted on my behalf to the State Water Board's GeoTracker website." This letter must be signed by the Responsible Party, or legally authorized representative of the Responsible Party.

PROFESSIONAL CERTIFICATION & CONCLUSIONS/RECOMMENDATIONS

The California Business and Professions Code (Sections 6731, 6735, and 7835) requires that work plans and technical or implementation reports containing geologic or engineering evaluations and/or judgments be performed under the direction of an appropriately licensed or certified professional and include the professional registration stamp, signature, and statement of professional certification. Additional information is available on the Board of Professional Engineers, Land Surveyors, and Geologists website at: http://www.bpelsg.ca.gov/laws/index.shtml.

UNDERGROUND STORAGE TANK CLEANUP FUND

For LUFT cases, RP's non-compliance with these regulations may result in ineligibility to receive grant money from the state's Underground Storage Tank Cleanup Fund (Senate Bill 2004) to reimburse the cost of cleanup. Additional information is available on the internet at: <u>https://www.waterboards.ca.gov/water_issues/programs/ustcf/</u>

AGENCY OVERSIGHT

Significant delays in conducting site assessment/cleanup or report submittals may result in referral of the case to the Regional Water Board or other appropriate agency, including the County District Attorney, for possible enforcement actions. California Health and Safety Code, Section 25299.76 authorizes enforcement including administrative action or monetary penalties of up to \$10,000 per day for each day of violation.